f

Geochemistry, Geophysics, Geosystems[®]

RESEARCH ARTICLE 10.1029/2023GC011118

Key Points:

- Anisotropic receiver function analysis reveals multiple anisotropic layers in the crust and upper mantle beneath southern New England
- Anisotropic geometries at <10 km depth resolved by Bayesian inversion imply contributions from sediment layering and Paleozoic deformation
- Array-wide anisotropic interfaces in the upper mantle show plate motion parallel flow distorted by localized mantle upwelling

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

Y. Luo, yantao.luo@yale.edu

Citation:

Luo, Y., Long, M. D., Link, F., Karabinos, P., & Kuiper, Y. D. (2023). Insights from layered anisotropy beneath southern New England: From ancient tectonism to present-day mantle flow. *Geochemistry, Geophysics, Geosystems,* 24, e2023GC011118. https://doi. org/10.1029/2023GC011118

Received 10 JUL 2023 Accepted 28 NOV 2023

Author Contributions:

Conceptualization: Yantao Luo, Maureen D. Long Data curation: Yantao Luo Formal analysis: Yantao Luo, Maureen D. Long, Frederik Link, Paul Karabinos, Yvette D. Kuiper Funding acquisition: Maureen D. Long Investigation: Yantao Luo, Maureen D. Long, Frederik Link, Paul Karabinos, Yvette D. Kuiper Methodology: Yantao Luo, Frederik Link

© 2023 The Authors. Geochemistry, Geophysics, Geosystems published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

LUO ET AL.

Insights From Layered Anisotropy Beneath Southern New England: From Ancient Tectonism to Present-Day Mantle Flow

Yantao Luo¹ , Maureen D. Long¹, Frederik Link¹, Paul Karabinos², and Yvette D. Kuiper³

¹Department of Earth and Planetary Sciences, Yale University, New Haven, CT, USA, ²Department of Geosciences, Williams College, Williamstown, MA, USA, ³Department of Geology and Geological Engineering, Colorado School of Mines, Golden, CO, USA

Abstract Seismic anisotropy beneath eastern North America, as expressed in shear wave splitting observations, has been attributed to plate motion-parallel shear in the asthenosphere, resulting in fast axes aligned with the plate motion. However, deviations of fast axes from plate motion directions are observed near major tectonic boundaries of the Appalachians, indicating contributions from lithospheric anisotropy associated with past tectonic processes. In this study, we conduct anisotropic receiver function (RF) analysis using data from a dense seismic array traversing the New England Appalachians in Connecticut to examine anisotropic layers in the crust and upper mantle and correlate them with past tectonic processes as well as present-day mantle flow. We use the harmonic decomposition method to separate directionally-dependent variations of RFs and focus on features with the same harmonic signals observed across multiple stations. Within the crust, there are multiple features that may be correlated with stratification in the Hartford Basin, faults in the Taconic thrust belt, shear zones formed during Salinic/Acadian terrane accretion events, and orogen-parallel crustal flow in the Acadian orogenic plateau. We apply a Bayesian inversion method to obtain quantitative constraints on the direction and strength of intra-crustal anisotropy beneath the Hartford Basin. In the upper mantle, we identify a fossil shear zone possibly formed during oblique subduction of Rheic Ocean lithosphere. We also find evidence for a plate motion-parallel flow zone in the asthenosphere that is likely disturbed by mantle upwelling near the southern margin of the Northern Appalachian Anomaly in the eastern part of the study area.

Plain Language Summary Seismic waves with different propagation and oscillation directions can exhibit different velocities when going through a medium with some directional properties; this phenomenon is called seismic anisotropy. Seismic anisotropy observed beneath eastern North America is often attributed to present-day flow in the upper mantle. The mantle flow causes shear waves oscillating in the direction of flow (e.g., in the direction of North America plate motion) to travel faster than those that travel in other directions. However, this pattern does not hold true for some regions in the Appalachian Mountains, suggesting that past tectonic events can result in long-lived, "frozen-in" anisotropy in the lithosphere, which modifies the predicted anisotropic behavior beneath these regions. In this study, we investigate sources of seismic anisotropy beneath southern New England using a method based on directionally dependent variations of P-wave to S-wave conversions at interfaces with contrasts in anisotropy. This method can separate signals caused by different anisotropic features and constrain the depth distribution of anisotropy. We identify various anisotropic layers in the crust and upper mantle and correlate them with Appalachian mountain-building events and complex present-day flow in the upper mantle.

1. Introduction

Seismic anisotropy, the directional dependence of seismic wave velocities, has been widely observed in Earth's crust and upper mantle (e.g., Babuska & Cara, 1991; Gung et al., 2003; Park & Levin, 2002). Seismic anisotropy in the crust and upper mantle can have many potential causes, which can generally be classified into either shape preferred orientation or lattice/crystallographic preferred orientation (CPO) of Earth materials (Fouch & Rondenay, 2006). In the Earth's shallow crust, seismic anisotropy can be attributed to the alignment of cracks and pores (Chapman, 2003; Crampin, 1981; Kaneshima et al., 1988); in the deeper crust, ductile deformation of metamorphic rocks resulting in foliation and CPO of crustal minerals like amphibole, mica, or quartz can cause anisotropy (Barruol & Mainprice, 1993; Christensen & Mooney, 1995; Mainprice & Nicolas, 1989). Upper mantle anisotropy is usually attributed to the CPO of olivine (Hansen et al., 2021; Karato, 2008; Karato & Wu, 1993), but in some cases it may be complicated by the presence of segregated melts (Holtzman et al., 2003).





Project Administration: Maureen D. Long

Resources: Maureen D. Long Software: Yantao Luo, Frederik Link Supervision: Maureen D. Long Validation: Yantao Luo, Frederik Link Visualization: Yantao Luo, Frederik Link Writing – original draft: Yantao Luo Writing – review & editing: Yantao Luo, Maureen D. Long, Frederik Link, Paul Karabinos, Yvette D. Kuiper of seismic anisotropy can indicate ongoing processes, such as present-day mantle flow in the asthenosphere (e.g., Gung et al., 2003; Lopes et al., 2020), as well as relict structures in the lithosphere from past tectonic events (e.g., Hopper et al., 2017; Long et al., 2017).

The eastern margin of North America has undergone two complete supercontinental cycles, including the Proterozoic Grenville orogenic cycle and the Paleozoic Appalachian orogenic cycle. The Grenville orogenic cycle involved the Mesoproterozoic formation of the supercontinent Rodinia and subsequent Neoproterozoic continental rifting (Bogdanova et al., 2009; Rivers, 1997). The Appalachian orogenic cycle started with diachronous accretion of various Gondwana-derived terranes to the Laurentian margin during the Ordovician Taconic orogeny, the Late Ordovician-Early Silurian Salinic orogeny, the Latest Silurian-Devonian Acadian orogeny, and the Late Devonian Neoacadian orogeny (e.g., Hatcher, 2010; Hibbard et al., 2007, 2010; van Staal et al., 2009). Appalachian orogenesis culminated in the continent-continent collision of composite Laurentia with Gondwana and the formation of the supercontinent Pangea during the Late Mississippian to Permian Alleghanian orogeny (Hatcher, 2002), followed by Mesozoic continental rifting and the breakup of Pangea (Withjack et al., 2012). Multiple episodes of subduction, terrane accretion, continental collision, and rifting involved in these two supercontinent cycles greatly modified both the surface geology and the lithospheric structure of eastern North America. These tectonic processes, while being transient events in the context of Earth's vast geologic timeline, have likely deformed the lithosphere and potentially left a long-lasting anisotropic structure that can be investigated today by seismic techniques.

Shear wave splitting analysis, based on measurements of the splitting or birefringence of shear waves, is a commonly-used and powerful technique to study seismic anisotropy, particularly in the mantle (Long & Silver, 2009; Savage, 1999). Splitting analysis has previously been applied to eastern North America at different scales (e.g., Aragon et al., 2017; Fouch et al., 2000; Link et al., 2022; Long et al., 2016; Lopes et al., 2020; Yang et al., 2017). Long et al. (2016) conducted SKS splitting analysis on USArray Transportable Array stations across eastern North America and found imperfect correlations between the fast axis direction and absolute plate motion (APM). Notably, Long et al. (2016) and White-Gaynor and Nyblade (2017) pointed out an orogen-parallel fast direction pattern beneath the Appalachian orogen, which suggested substantial contributions from crust and lithospheric mantle anisotropy. In contrast, Yang et al. (2017) conducted a shear wave splitting study at a similar scale and proposed a more APM-parallel overall fast direction beneath eastern North America, with little contribution from lithospheric deformation associated with past tectonic events. Applying the SKS splitting technique to a more densely spaced seismic array in the central Appalachians, Aragon et al. (2017) confirmed the presence of an orogen-parallel fast direction pattern beneath the Appalachians, with a sharp lateral transition in splitting behavior along the array, which they attributed to likely contributions from the lithosphere. To the north, Lopes et al. (2020) found generally APM-parallel fast directions beneath southern New England, but found that the splitting delay times decreased drastically from west to east. They suggested that weaker splitting beneath the eastern portion of the study area may result from either destructive interference from contributions from the lithosphere or from a change in the flow direction in the asthenosphere, possibly associated with the presence of the Northern Appalachian Anomaly (NAA), a prominent low velocity anomaly centered beneath New England (e.g., Levin et al., 1995, 2018; Li et al., 2003; Tao et al., 2020). The ambiguity in the interpretations involved in shear wave splitting studies is rooted in the fact that shear wave splitting is a path-integrated method, so it intrinsically lacks the vertical resolution and it cannot uniquely distinguish contributions from lithosphere and asthenosphere anisotropy (Park & Levin, 2002).

One way to complement shear wave splitting observations and obtain constraints on the vertical distribution of anisotropy is via anisotropic Ps receiver function (RF) analysis, which is based on the observation of directionally dependent variations of Ps converted phases (Frederiksen & Bostock, 2000; Levin & Park, 1997; Park & Levin, 2016a). Anisotropy-aware RF analysis has been applied to both subduction zones (e.g., Bar et al., 2019; Haws et al., 2023; Wirth & Long, 2014) and continental settings (e.g., Ford et al., 2016; Schulte-Pelkum & Mahan, 2014b; Wirth & Long, 2014) to investigate potential multi-layered anisotropy beneath those study regions. This technique has also been applied to eastern North America (e.g., Frothingham et al., 2022; Li et al., 2021; Long et al., 2017; Yuan & Levin, 2014). Yuan and Levin (2014) conducted both SKS splitting and the anisotropic RF analyses on three long-running stations in northeastern US, and found coherent lower lithosphere anisotropy with a fast axis nearly orthogonal to the strike of the Appalachian orogen and APM-consistent asthenosphere anisotropy. Long et al. (2017) compared anisotropic RFs from stations in the Grenville Province and in the Appalachian domains and identified dissimilar anisotropic features in the crust and lithospheric mantle, which suggested that different tectonic processes had modified the lithosphere beneath these two regions. Li et al. (2021) examined RFs of 62 stations in northeastern North America, focusing on discontinuities in the upper mantle. They identified multiple negative velocity gradients accompanied by contrasts in anisotropy in the lithosphere at 60–100 km depth beneath most stations, and they found the lithosphere to be thinnest beneath central New England, where the NAA is centered. Despite the valuable insights on anisotropic layering beneath individual stations gleaned from these studies, their interpretations have been hindered by an inability to establish clear correlations between the features observed beneath each station. As a result, a comprehensive investigation of any given feature becomes exceedingly challenging. This limitation can be partially attributed to the relatively large distance between the seismic stations used in these studies. It is difficult to trace localized structures such as shear zones and faults across stations that are more than 70 km apart (the average spacing of TA stations), especially because the strike and dip of those features are not known, and structures vary abruptly perpendicular and parallel to the orogen. This problem can be mitigated by using a denser seismic array, which potentially allows the identification of individual features across several stations, thus providing constraints on the shape, orientation, and lateral extent of anisotropic features, and aiding the interpretation of their nature and origin.

The Seismic Experiment for Imaging Structure beneath Connecticut array (SEISConn; Long & Aragon, 2020), a dense broadband seismic array in southern New England, provides an opportunity to use the anisotropic RF technique to investigate anisotropic structures beneath this region in detail. The presence of multiple layers of anisotropy beneath the SEISConn array was proposed in a previous SKS splitting study (Lopes et al., 2020) as well as a previous RF study focusing on radial component seismograms (Luo et al., 2021). In this study, we unravel the anisotropic complexity beneath southern New England in detail by analyzing both radial and transverse components of RFs, performing harmonic decomposition to extract backazimuth-dependent terms, and conducting Bayesian inversion to resolve detailed anisotropic geometries and explore potential interpretations. We identify several anisotropic interfaces in the crust and lithospheric mantle that can be traced beneath adjacent stations and that may represent relict shear zones or fault zones related to past tectonic events. Additionally, two deeper anisotropic interfaces at 80–160 km depths are consistently observed beneath the entire array, which may reflect a change in the dominant mantle flow direction beneath the study region. Findings from this detailed regional study can be extrapolated to other segments of the Appalachian orogen and to other collision zones to understand how orogenesis contributes to lithospheric architecture and how deformation may be recorded by seismic anisotropy. This study also demonstrates how these anisotropic features might be modified or overprinted by younger processes.

2. Data and Methods

2.1. The Seismic Experiment for Imaging Structure Beneath Connecticut (SEISConn)

The SEISConn array consisted of 15 broadband seismometers, named CS01 to CS15 from west to east, deployed with ~10 km spacing across northern Connecticut from 2015 to 2019 (Figure 1a; Long & Aragon, 2020). The array traverses the eastern edge of Laurentia near station CS05, which separates the exposed Proterozoic Grenville crust to the west and the Gondwana-derived Moretown terrane crust accreted during the Ordovician Taconic orogeny to the east. Stations CS06-CS09 in the central portion of the array are situated above the Hartford basin, which was opened by the Mesozoic rifting event. To the east, there is likely a buried suture associated with the accretion of Ganderia during the Late Ordovician-Early Silurian Salinic orogeny. Farther east, near station CS14, the array traverses the exposed western edge of Avalonia, another Gondwana-derived terrane that was accreted to the composite Laurentia during the Latest Silurian-Devonian Acadian orogeny.

The main goal of the SEISConn deployment is to understand how the crust and mantle lithosphere beneath southern New England have been modified by past orogenic cycles and younger processes through investigations of the seismic velocity structure, discontinuities, and anisotropy beneath this region. A detailed crustal velocity model has been constructed by Gao et al. (2020) using the full-wave ambient noise tomography technique. They found a low-velocity mid-crust, likely reflecting radial anisotropy produced during the Mesozoic rifting, and a high-velocity lower crust beneath the Hartford Basin, which might be related to magmatic underplating during the Central Atlantic Magmatic Province event. Traditional 1-D RF analysis using the converted Ps phase (Luo et al., 2021) showed a ~15 km Moho depth offset between station CS03 and CS04, as well as multiple discontinuity features in the crust and upper mantle that might be associated with subduction and terrane accretion events during the Appalachian orogenic cycle. A lithospheric step of similar size is also suggested beneath southern New England by RF analysis using the converted Sp phase (Goldhagen et al., 2022). A follow-up study using a more





Figure 1. (a) Station map of the SEISConn array, with red triangles denoting station locations. Light green and dark green crosses denote the locations of *P*-to-*S* conversions of all events used in this study, at 40 and 100 km depths, respectively. Colored lines delineate important geologic boundaries in this region, as indicated by the legend. Thin black lines represent the US state boundaries. CT—Connecticut, MA—Massachusetts, NY—New York, RI—Rhode Island. (b) Event map that includes all events used for the anisotropic RF analysis. The red triangle marks the location of the SEISConn array, and red stars mark the locations of earthquakes used. Four dashed black circles denote 20°, 30°, 90°, and 155° distances from the SEISConn array, respectively. The event selection procedure is included in Text S1 of the Supporting Information S1.

advanced 2-D scattered wavefield migration technique (Luo et al., 2022) revealed a potentially doubled Moho at the Moho offset, which suggests the reactivation of the Proterozoic rift fault and overthrusting of rifted Grenville crust from the east onto the undeformed Grenville crust in the west. Luo et al. (2022) also further constrained the presence of a relict slab in the lithosphere, possibly subducted during the closure of the Rheic Ocean, and a localized low velocity anomaly that might be associated with the NAA in the upper mantle. The potential presence of the NAA beneath southeastern New England was also implied by SKS splitting analysis (Lopes et al., 2020), which showed a generally APM-parallel fast axis orientation with a gradually decreasing splitting delay time toward the east, near the likely southern margin of the NAA. However, the potentially multi-layered upper mantle anisotropy inferred in this study was not explicitly modeled by applying a grid-search and curve-fitting technique to the splitting results, due to insufficient backazimuthal coverage of SKS phases (Lopes et al., 2020). The scattered wave analyses (Luo et al., 2021, 2022) focused on layers with isotropic velocity contrast, and the SKS splitting study (Lopes et al., 2020) focused on the path-integrated signature of upper mantle anisotropy beneath southern New England. Although multiple layers of anisotropy with varying geometries were suggested by several of those studies, the depths and characteristics of individual anisotropic layers could not be explicitly resolved. This knowledge gap can be filled with the anisotropy-aware RF analysis that we carried out here.

2.2. Anisotropic Ps Receiver Function (RF) Analysis and Harmonic Decomposition

Ps receiver function analysis is based on identifying S-waves converted from incident P-waves at structural discontinuities with changes in material properties (Langston, 1979; Rondenay, 2009). We rotated seismograms to the *L-Q-T* coordinate system, where *L* is approximately parallel to the direct P wave, *Q* is orthogonal to *L* in the vertical plane, and *T* is orthogonal to both *L* and *Q*. The radial and transverse components of RFs are calculated by deconvolving the *L* component from the *Q* and *T* components of seismograms, respectively, using the multiple-taper-correlation method (Park & Levin, 2000, 2016b). (Note that for simplicity, we use the more conventional term "radial RF" for the RFs calculated based on the *Q*-component, as in Ford et al. (2016)). A detailed description of our data processing procedures and quality control criteria is in Text S1 of the Supporting Information S1. As shown in Figure 1b, earthquakes used as sources are found at a wide range of backazimuths, resulting in incident waves arriving from a range of directions.

When the seismic structure beneath a receiver is made up of horizontal layers with only isotropic velocity contrasts (Figure 2a), the *P*-to-*S* conversion will only produce vertically polarized SV waves whose amplitudes do not vary with backazimuth. However, when the interface is dipping (Figure 2b) or when one of the layers is anisotropic





Figure 2. Schematic diagrams demonstrating the variation of *P*-to-SV conversion amplitudes in different layer setups. Red lines denote incident P-waves, and blue lines denote converted SV waves. (a) Both the top and the bottom layers are isotropic, and the interface in between is flat. (b) Both layers are isotropic, but the interface in between is west-dipping. (c) The bottom layer has an east-plunging fast axis anisotropy. (d) The bottom layer has E-W oriented horizontal fast axis anisotropy.

with a plunging (Figure 2c) or horizontal (Figure 2d) axis of symmetry, both SV and horizontally polarized SH waves are produced, and their polarities and/or amplitudes vary systematically with backazimuth (Levin & Park, 1997). Specifically, if a slow layer overlies a fast layer across a west-dipping interface (Figure 2b), the *P*-to-SV conversion will be stronger for the incident wave coming from the west and weaker for the incident wave coming from the east. If the interface is horizontal but the lower layer is anisotropic with an east-plunging fast axis (Figure 2c), the *P*-to-SV conversion will be stronger for the incident wave coming from the east and weaker for the incident wave coming from the west. If the fast axis in the lower layer is instead horizontal and oriented E-W (Figure 2d), the *P*-to-SV conversion will be stronger for the incident wave coming from the east or west and weaker for the incident wave coming from the north or south. For all the cases mentioned above, if the positions of the upper and lower layers are switched, the directional dependence of the *P*-to-SV conversion will also be inverted. The directionally-dependent *P*-to-SV conversion is always accompanied by directionally-dependent *P*-to-SH conversion will be structure is horizontally layered and can be fully modeled by transverse isotropy (TI) with a cylindrical axis of symmetry (Park & Levin, 2016a).

Anisotropic RF analysis can reveal the presence of dipping interfaces and anisotropic layers by exploring the event backazimuth-dependent variations in the radial component RFs, which record the *P*-to-SV conversion, and the transverse component RFs, which record the *P*-to-SH conversion. Of course, the Earth structure is usually much more complicated than the two-layer examples shown in Figure 2. There can be multiple anisotropic layers with different anisotropic geometries separated by dipping interfaces, and the anisotropy involved can be more complicated than TI. As a result, it is often difficult to analyze the backazimuth-dependent variations of RFs directly because the signals generated from different features are blended. One way to more clearly characterize dipping interfaces and anisotropic orientations is to apply the harmonic decomposition technique (Bianchi et al., 2010; Olugboji et al., 2016; Park & Levin, 2016a; Shiomi & Park, 2008). This technique uses linear regression to model the behavior of both the radial and transverse RFs, extracting the constant term (the zeroth-order harmonic term) that does not vary with backazimuth as well as the higher-order harmonic terms (e.g., sin(baz), cos(baz), sin(2*baz)) that do vary with backazimuth.

Figure 3 provides three synthetic examples to demonstrate the capabilities of harmonic decomposition. The first model (Figure 3a) consists of two layers over a half-space, with a 30-km thick isotropic upper crust and a 5-km





Figure 3. Synthetic RF examples demonstrating the harmonic decomposition method for a range of isotropic and anisotropic scenarios. (a–c) Sketch of seismic models used to generate synthetic RFs using PyRaysum (Bloch & Audet, 2023; Frederiksen & Bostock, 2000). (d–f) Radial and transverse components of synthetic RFs, shown as a function of backazimuth (*y*-axis). Data are band-passed between 0.02 and 1.5 Hz. Positive signals are shown in red, and negative signals are shown in blue. The *x*-axis is expressed in units of time (sec), with the solid green line marking 0 km depth, and each subsequent green dashed line marking 10-km increment in depth, as calculated from the velocity model shown in (a)–(c) and assuming vertical incidence. (g–i) Constant and harmonic terms decomposed from calculated RFs. RFs are migrated to depth using the velocity model shown in (a)–(c) before harmonic decomposition. The "negative depth" does not have a physical meaning; it displays signals before the 0 s delay time in the time domain RF, which can be used as a benchmark for the overall noise level for each harmonically decomposed RF term. Unmodeled terms that are inconsistent with the assumptions of lateral homogeneity and TI anisotropy are shown in Figure S1 of the Supporting Information S1.

thick anisotropic lower crust that has the same isotropic velocity as the upper crust, but with an east-plunging +10% fast axis of anisotropy. The Moho is at 35 km depth, and the isotropic mantle half-space has faster seismic velocities than the crustal layers above. We calculate synthetic RFs using the PyRaysum program (Bloch & Audet, 2023), which generates ray-theoretical seismograms for incident plane waves propagating through seismic models with dipping interfaces and layered anisotropy (Frederiksen & Bostock, 2000). Figure 3d displays the calculated synthetic radial and transverse RFs in backazimuth gathers, with trivial random noise added. There are clear backazimuth-dependent variations of both radial and transverse component RFs at arrival times that correspond to 30 and 35 km depth, representing the top and bottom interfaces of the anisotropic lower crust. The variation in arrivals at 30 and 35 km depth on the radial component can be modeled as first-order positive and negative sine functions of backazimuth, respectively, and the patterns in the transverse component are correspondingly $+90^{\circ}$ phase shifted. These patterns can be extracted using the harmonic decomposition technique and are visible in the sin(baz) harmonic term (Figure 3g). The isotropic velocity increase across the Moho is visible

in the constant term at 35 km depth, as expected. The harmonic decomposition also shows that the east-plunging anisotropic axis also produces some minor cos(2*baz) signals in addition to the prominent sin(baz) signals. The synthetic RFs are computed based on a horizontally layered TI model, and therefore no significant energy is present in the unmodeled portion of RFs (Figure S1a in Supporting Information S1). Similarly, Figures 3e and 3h show the synthetic RFs in backazimuth gathers and the corresponding harmonic decomposition, respectively, calculated from a seismic model with an anisotropic lower crust with a horizontal +10% fast axis of anisotropy (Figure 3b). A completely isotropic seismic model with a dipping interface (Figure 3c) can also produce backazimuth-dependent variation in RFs (Figure 3f), which can be modeled as harmonic terms (Figure 3i). The presence of a dipping interface can cause backazimuth-dependent variation in the direct P phase, which manifests as a substantial sin(baz) signal at 0 s delay time in Figure 3i. In addition, there are larger unmodeled harmonic signals associated with the dipping interface compared with a model with anisotropy in horizontal layers (Figure S1).

2.3. Receiver Function Forward Modeling and Bayesian Inversion

For the synthetic example shown in Figure 3, we can qualitatively infer the potential anisotropic geometries and dipping interfaces involved in the seismic model (Figures 3a-3c) by visually inspecting the harmonic terms (Figures 3g-3i). In the case of particularly prominent anisotropic features that have significantly larger amplitudes than other signals, it is feasible to derive quantitative constraints on the strength and orientation of the anisotropy. One way to do this is via forward modeling, in which we fit the observed harmonic terms with synthetic RFs calculated for models with parameters based on a combination of prior knowledge and trial-and-error (e.g., Bianchi et al., 2010; Ford et al., 2016; Liu et al., 2015). With many unknown parameters, such forward modeling usually involves a substantial amount of trial-and-error, and it is often difficult to fully explore the tradeoffs among parameters such as the dipping angle of the symmetry axis and the strength of the anisotropy.

These difficulties can be mitigated by applying Bayesian inversion, which infers values of model parameters from a probabilistic perspective, instead of doing extensive trial-and-error forward modeling. A Markov chain Monte Carlo approach has been applied in several previous RF studies (e.g., Bissig et al., 2021; Bodin et al., 2013, 2016; Wirth et al., 2016). Here, we use a Bayesian framework with efficient ray-theoretical forward calculations (Link et al., 2020) to invert the anisotropic and dipping interface model that best fits the observed constant and harmonic terms. One advantage of fitting the harmonically decomposed RFs, instead of traditional RFs or raw seismograms, relates to the fact that the harmonic terms are extracted based on the assumption of lateral homogeneity and TI anisotropy with a cylindrical axis of symmetry; therefore, patterns of signals that cannot be produced by dipping interfaces and TI anisotropy (e.g., scattering due to lateral heterogeneity, effects from varied incident wave slowness due to different epicentral distances, anisotropy that cannot be approximated using a cylindrical axis of symmetry) are suppressed. Therefore, applying Bayesian inversion to the harmonically decomposed RFs has the potential to infer the anisotropic seismic model under the TI assumption more faithfully, without attempting to fit other unrelated signals and artifacts.

3. Results

We first present results from a single station of the SEISConn array, station CS07, both as a case study that illustrates how we apply our analysis to real data, and as an example of a station that exhibits strong, clear signals from crustal anisotropy that can be modeled using our Bayesian inversion technique. Next, we present results from across the SEISConn array, with a focus on identifying signals that are laterally coherent across multiple stations.

3.1. Case Study: CS07

CS07 is a station deployed in the Hartford Basin in the central part of the SEISConn array (Figure 4a). The station has good event backazimuth coverage (Figure 4b), with 149 events selected for RF analysis. On the radial component, the Moho conversion is visible as a positive signal at 3.5-4.5 s that is coherent across all backazimuths. Perhaps a more eye-catching feature is a set of signals at 0.5-1.0 s whose amplitudes and polarities vary with backazimuth. There is a negative signal followed by a positive signal at 0.5-1.0 s in the $45^{\circ}-145^{\circ}$ backazimuth range, which is reversed to a positive signal followed by a negative signal at $185^{\circ}-355^{\circ}$ backazimuth range. Signals at the backazimuth range in between (i.e., $145^{\circ}-185^{\circ}$ and $355^{\circ}-45^{\circ}$) have weaker amplitudes





Figure 4. Results from the analysis at station CS07. (a) Map of station location with plotting conventions as in Figure 1a. (b) Radial and transverse component RFs shown as backazimuth gathers. Data are band-passed between 0.02 and 1.5 Hz. Plotting conventions are as in Figure 3d. (c) Constant and harmonic terms decomposed from CS07 RFs. RFs are migrated to depth using a modified version of the IASP91 velocity model (Table S1 in Supporting Information S1; Kennett & Engdahl, 1991). Unmodeled terms are shown in Figure S1d of the Supporting Information S1.

at 0.5–1.0 s. This backazimuthally dependent pattern also manifests in the transverse component with a $+90^{\circ}$ phase shift, such that the negative followed by positive signals occur at 135°–235° while the positive followed by negative signals occur at 275° -85°. This correlation between the radial and the transverse RFs suggests that this backazimuth-dependent variation pattern can be generated by TI anisotropy. Indeed, this feature is identified with prominent sin(baz) signals in the harmonic decomposition (Figure 4c). We observe a large negative signal at \sim 3 km depth followed by another large positive signal at \sim 6 km depth in the sin(baz) term; these likely represent the sharp top and bottom interface of a \sim 3 km thick anisotropic layer. The pattern in the sin(baz) term can be explained by either a west-plunging fast axis or an east-plunging slow axis of symmetry in the anisotropic layer. This ambiguity is typically difficult to resolve in practice; the difference between the two cases lies in the smaller-amplitude accompanying signals on the cos(2*baz) term. Specifically, a west-plunging fast axis will yield negative followed by positive signals, but an east-plunging slow axis will yield positive followed by negative signals. Uncommonly, the feature we observe beneath CS07 is prominent enough that the accompanying signals in the cos(2*baz) term can be recognized as positive followed by negative at the corresponding depths. Therefore, this anisotropic layer likely has an east-plunging slow axis of symmetry. Above this anisotropic layer, there is a slower layer, indicated by a positive signal at \sim 3 km depth in the constant term, which represents an isotropic velocity increase with depth. This slow layer likely represents the sedimentary units of the Hartford Basin, given the location of CS07 and the depth of the interface. The Moho also shows up clearly at ~ 26 km depth in the constant term, as expected. Minor signals are present in the unmodeled RFs (Figure S1d in Supporting Information S1), which can arise from effects such as lateral heterogeneities, noise, and irregular event distribution. RF results for all individual SEISConn stations, similar to the example from the CS07 shown in Figure 4, are shown in Supporting Information S1 (Figures S2-S15).

We take advantage of the large amplitude of the conversions associated with the shallow crustal anisotropic layer beneath CS07 and apply Bayesian inversion to obtain more quantitative constraints on its anisotropic geometry as well as its dipping top and bottom interfaces (Figure 5). The inversion is conducted on the harmonic terms derived from RFs that are band-passed filtered between 0.02 and 1.5 Hz because signals from the shallow crustal

8 of 23





Figure 5. Results of Bayesian inversion for crustal structure beneath station CS07. (a) Sketch (not to scale, but with layer thicknesses (H) listed at right) of the most likely model for crustal anisotropy beneath CS07, as constrained by the Bayesian inversion, showing the mean value of each parameter. Statistical distributions of parameters are displayed in Figures S16–S23 of the Supporting Information S1. (b) Back azimuthal gathers of synthetic RFs calculated from the mean Bayesian model shown in (a). The plotting convention is the same as in Figure 4b. (c) Comparison of Bayesian inverted harmonic terms with the observation. The black lines represent the observed harmonic RFs for CS07, as in Figure 4c, but in time domain. The orange lines show the harmonic RFs calculated from the mean value of each parameter constrained by the Bayesian inversion shown in (a). The red lines show the harmonic RFs calculated from the mean value of each parameter constrained by the Bayesian inversion. The vertical solid green line marking 0 km depth, and each subsequent green dashed line marking 10-km increment in depth, as calculated from the velocity model shown in Table S1 of the Supporting Information S1 and assuming vertical incidence.

interfaces can be well isolated with this frequency band, as shown in Figure 4. An overview of the parameter space setup is shown in Table 1, and more details of the inversion procedure are available in Supporting Information S1. The mean values of the Bayesian statistics (Figure 5a) can reconstruct synthetic RFs that capture all major crustal signals observed in the raw RFs of CS07, as evident in the resemblance between Figures 4b

Table 1

Model Parameters for Our Bayesian Inversion Searching for a Three-Layered Anisotropic Model With Layers 1–3 From the Surface to a Halfspace at Depth, Considering Dipping Interfaces

	H in (km)		v_p in (km/s)	v_P/v_S		ρ in (g/cm ³)	<i>a</i> in (%)		<i>φ</i> in (°)		θ in (°)		γ in (°)		δ in (°)		
Lower/upper bo	ounds																
Layer 1	1.4	8.6	5	1.85	3	2.6	-25	0	0	180	-60	60	0	360	0	30	
Layer 2	1.1	6.8	5.8	1.5	3	2.8	-25	0	0	180	-60	60	0	360	0	30	
Layer 3	6.2	37	5.8	1.5	3	2.8	-5	0	0	180	-60	60	0	360	0	30	
Initial values																	
Layer 1	2.5		Fixed	2.0		Fixed	Random		Random		0		0		0		
Layer 2	5.3		Fixed	1.73		Fixed	Random		Random		0		0			0	
Layer 3	21.7		Fixed	1.73		Fixed	Random		Random		0		0			0	

Note. The parameters for P-wave velocity and density ($\mathbf{v}_{\mathbf{p}}$ and ρ) are kept fixed in all inversion runs due to limited sensitivity for these parameters. Thickness (*H*), P-wave to S-wave velocity ratio ($\mathbf{v}_{\mathbf{p}}/\mathbf{v}_{\mathbf{s}}$), anisotropic strength (*a*), the trend of slow axis from north (ϕ), its plunge from the horizontal (θ), the interface dip (δ), and the strike of the layer interfaces (γ) are parameters which are searched for in the inversion and vary between the boundaries separated by the dashed lines. Bayesian inversion is performed for several different initial starting models, where the strength of anisotropy and fast axis direction are chosen randomly from the given interval, while the remaining free parameters are set to the parameters given in the lower half of the table.

and 5b. When comparing the harmonically decomposed RFs, the Bayesian inversion provides an excellent fit of the prominent $\sin(baz)$ signals and closely matches the signals in other terms (Figure 5c). The robustness of the inversion can be assessed by the statistical distribution of each inverted parameter (Figures S16–S23 in Supporting Information S1). The inversion resolves an east-plunging slow axis in the second layer with >20% anisotropy situated between a slower top layer and a (nearly) isotropic deeper, third layer with similar velocities, consistent with our qualitative interpretation based on visual inspection. Furthermore, the inversion sheds light on the potential existence of a west-plunging slow axis with ~10% anisotropy in the top layer, associated with the sedimentary units of the Hartford Basin, which was not as obvious from visual assessment. The more general application of such quantitative analysis. For example, the difference between an east-plunging slow axis and a west-plunging fast axis lies in very minor signals in the cos(2*baz) term, which are likely buried in noise and undistinguishable in typical cases. In the following analysis of the entire array, in which most features do not have as significant amplitudes as the shallow crustal features observed beneath CS07, we apply a more qualitative approach to interpretation.

3.2. Application to the Full SEISConn Array

Figure 6 shows the compiled results of harmonic analysis applied to data from the entire SEISConn array using a bandpass filter between 0.02 and 0.1 Hz. The constant RF profile is generally consistent with results from previous isotropic analyses (Luo et al., 2021, 2022), showing a large Moho depth offset between stations CS03 and CS04. In the profiles for the higher-order terms of the harmonic analysis, we can observe a plethora of signals with varied amplitudes and polarities, indicating an abundance of sharp anisotropic contrasts and/or dipping interfaces beneath this region. In general, harmonic signals in the crust have larger amplitudes than those at greater depths, and harmonic signals beneath the western portion of the array have larger amplitudes than those beneath the eastern portion of the array. We can identify several features, both in the crust and in the upper mantle, that have coherent harmonic signals across several adjacent stations.

In order to first identify signals within the crust and shallowest mantle, we show in Figure 7 a set of profiles similar to those in Figure 6 but focused on the upper 80-km depth range beneath the study region and with major anisotropic and/or dipping features labeled. Data used to construct these images were bandpass filtered between 0.02 and 1.5 Hz to emphasize shallower, smaller-scale features. The largest harmonic signals are associated with Feature #1 in the sin(baz) profile (Figure 7). This feature is identifiable beneath stations CS05 to CS09, with larger amplitudes beneath CS05, CS06, and CS07 and smaller amplitudes beneath CS08 and CS09. As discussed in Section 3.1, this negative followed by a positive signal pattern in the sin(baz) term likely denotes an anisotropic layer in the shallow crust with east-plunging slow axis anisotropy (Figure 5). This feature displays a zig-zag shape in cross section, and the top interface of this anisotropic layer aligns with the bottom of the Harford basin, except for the down-going branch beneath CS05. Feature #2 consists of positive signals near or before 0 s time and negative signals at shallow depths in the sin(baz) terms beneath CS01, CS02, and CS03. The signal around or even before 0 s delay time in harmonic RFs indicates backazimuth-dependent variation of the direct P arrival amplitude, which suggests the presence of dipping interfaces (Figure 3i; Schulte-Pelkum & Mahan, 2014a). In addition to the signal at or near 0 s delay time, a dipping interface will also result in the variation of P-to-S conversion amplitude at the depth of the interface, with opposite polarity to the variation of direct P. Therefore, Feature #2 likely represents a series of east- or west-dipping interfaces at 0-5 km depth beneath CS01, CS02, and CS03. Feature #3 is an elongated west-dipping interface that expresses itself in the cos(baz) term, which suggests a generally north- or south-plunging anisotropic axis. There is no accompanying interface with opposite polarity in the cos(baz) term, which implies that this interface can be either the top or the bottom interface of an anisotropic layer, with the other interface being diffuse and thus not observable by RF analysis. Features #4 and #5 in the sin(baz) profile align with the Moho geometry shown in the constant RF profile. The negative and positive sin(baz) signals are likely produced by the west- (in the west) and east- (in the east) dipping Moho discontinuities. In the cos(baz) term, we also observe positive signals associated with Feature #4 and negative signals associated with Feature #5; these provide information about the Moho geometry in the north-south direction, outside the profile plane. The character of these signals suggests that the Moho beneath the western portion of the profile is northwest-dipping, whereas the Moho beneath the eastern portion of the profile is southeast-dipping. Feature #6 is a west-dipping interface with a positive velocity gradient in the upper mantle that only shows up in the constant RF as a series of minor positive signals from ~56 km depth beneath CS15 to ~77 km depth beneath CS06. This





Figure 6. Harmonically decomposed RF profiles for the SEISConn array, with data band-passed between 0.02 and 1.0 Hz. The RFs migrate to depth based on a 1-D velocity model modified from IASP91 (Table S1 in Supporting Information S1; Kennett & Engdahl, 1991). For each trace, the black curve represents the bootstrap mean, and the pale green band shows ± 1 standard deviation range of the trace derived from the bootstrap test. Positive and negative regions under the curve that are beyond one standard deviation are filled with red and blue, respectively. The top left panel shows the constant RF profile, and the panels at the middle and to the right show harmonic RF profiles for higher-order terms, as indicated by labels at the top. The spacing between traces reflects the physical distances between stations along the profile in kilometers. The aspect ratio of the plots in this figure is 1:1. Unmodeled terms are shown in Figure S24 of the Supporting Information S1.

isotropic feature is also observed in the previous isotropic RF and scattered wavefield migration studies (Luo et al., 2021, 2022).

In order to identify features deeper in the mantle, we show in Figure 8 profiles similar to those in Figure 6 with data filtered at lower frequencies (0.02–0.5 Hz); in this view, deeper and broader structures come into focus. Feature #7 represents an extensive anisotropic layer beneath the Moho with a generally NE-SW fast or NW-SE slow horizontal axis of symmetry, indicated by a series of positive signals followed by negative signals in the sin(2*baz) term. Also, the presence of positive cos(baz) signals related to Feature #7 may imply varied anisotropic geometries within the inferred anisotropic layer. Deeper in the upper mantle, features #8 and #9 can be coherently observed beneath almost the entire array, and they are not artifacts associated with the arrival of multiple Moho reflection signals (magenta dashed lines in Figure 8). Feature #8 is a flat-lying feature observed at 80–100 km depth in the cos(baz) term, suggesting that it is either a north- or south-dipping interface, or an anisotropic interface associated with a north- or south-plunging anisotropic axis. Feature #9 is a west-dipping feature observed at 80–160 km depth in the sin(2*baz) term, associated with either a NE-SW fast or NW-SE slow horizontal anisotropic axis. The fact that these deep features are not as well imaged in higher frequencies (Figure 6) indicates that they represent more diffuse, gradational boundaries compared to





Figure 7. Annotated harmonically decomposed RF profiles for the SEISConn array, with data band-passed between 0.02 and 1.5 Hz to emphasize shallower and smaller-scale features. Features identified and discussed in this study are delineated using cyan dashed lines, and the feature number (#1–#6) is also annotated for each feature. The aspect ratio of the plots is 2:1, with the vertical axis exaggerated by a factor of 2. Other plotting conventions are as in Figure 6. A zoomed-in view of the crustal features (#1–#3) is shown in Figure S25 of the Supporting Information S1. Unmodeled terms are shown in Figure S26 of the Supporting Information S1.

the sharper boundaries observed in the crust. Also, the converted signals associated with these deep features are likely further modified and defocused when traveling through the shallower heterogeneous and anisotropic structures.

4. Discussion

Figure 9 shows the major features observed in the harmonic RFs along with their preferred interpretations. The signals appearing in the higher-order terms of harmonically decomposed RFs usually indicate the presence of a dipping interface, an interface with anisotropic contrast, or both (Park & Levin, 2016a). Because the extents and shapes of these features beneath the SEISConn array are constrained by the dense station spacing, we can use the RF variation patterns extracted by harmonic decomposition along with a priori knowledge about the geologic/ tectonic background to interpret specific crust and upper mantle structures (Figure 9). The set of features we document, with drastically different anisotropic geometries, is highly unlikely to have been formed by a single tectonic event. Instead, they likely reflect the varied deformation patterns of several different past tectonic events and/or present-day processes. In the following section, we will discuss in detail the potential origins and implications of the anisotropic and/or dipping interfaces beneath southern New England, and how the insights from this study can have wider applications and broader implications beyond this specific region.





Figure 8. Annotated harmonically decomposed RF profiles for the SEISConn array, with data band-passed between 0.02 and 0.5 Hz to emphasize deeper and larger-scale features. A thick magenta dashed line in the constant RF profile marks the expected arrival time of a hypothetical Moho multiple reflection. Features are labeled using the same system as in Figure 7. The aspect ratio of the plots is 1:1 with no vertical exaggeration. Other plotting conventions are as in Figures 6 and 7. Unmodeled terms are shown in Figure S27 of the Supporting Information S1.

4.1. Dipping Interfaces

Beneath stations CS01, CS02, and CS03 in the westernmost portion of the array, we observe a series of dipping interfaces at very shallow depths (0–5 km), identified as Feature #2 in the sin(baz) term (Figure 7). They can be either west- or east-dipping. Resolving this ambiguity solely based on the observed RFs proves to be challenging, given that these interfaces are so close to the surface and that the signals in the constant term can be severely affected by artifacts such as side-lobes of the incident P phase. The geographic location of this feature is on the Laurentian margin, to the west of the suture between Laurentia and the Gondwana-derived Moretown terrane accreted during the Ordovician Taconic orogeny (Karabinos et al., 2017; Macdonald et al., 2014). To the north of our study area, a previous seismic reflection study traversing eastern New York, Vermont, and New Hampshire (the COCORP line; Ando et al., 1984) revealed numerous east-dipping interfaces beneath the Laurentian margin in the New England Appalachians that may be interpreted as a system of Taconic thrust faults (Stanley & Ratcliffe, 1985). Feature #2 may then signify the presence of this east-dipping fault system beneath southern New England.

We find that the Moho beneath the SEISConn array is shallowest beneath the Hartford Basin in the central portion of the array, and it gradually deepens to the west and to the east, consistent with our previous results (Luo et al., 2021, 2022). Because of the large isotropic velocity contrast between the crust and the upper mantle,



Figure 9. East-west cross section diagram summarizing the features delineated in Figures 7 and 8. Black solid lines denote flat lying isotropic interfaces; cyan solid lines denote dipping isotropic interfaces; cyan dash-dotted lines denote interfaces that have both an isotropic seismic velocity contrast and an anisotropic contrast; cyan dotted lines denote interfaces that only have an anisotropic contrast. Arrows denote the inferred anisotropic fabric orientations or mantle flow directions. Crosses and circles denote in-and-out of the plane directions (north and south in this case). The sketch is not scaled to allow for the visibility of all features, but approximate depths of key interfaces are noted to the right.

the Moho manifests clearly in the constant RFs as positive signals (Figures 6-8). Furthermore, the west- and east-dipping portions of the Moho discontinuity result in negative and positive signals beneath the western and eastern portions of the array, respectively (Features #4 and #5). Feature #4, associated with the west-dipping Moho, may extend all the way to beneath CS02, beyond the prominent Moho depth offset located between CS03 and CS04 (Luo et al., 2021). The deeper Grenville Moho to the west of the offset is flat lying, depicted as a horizontal black line in Figure 9. Thus, unlike the west-dipping shallower Moho, the deeper Moho does not generate any coherent higher-order harmonic signals. In this way, the two distinct Moho interfaces can be clearly identified and distinguished beneath CS03 and CS02, which was not possible using our previous traditional RF analysis (Luo et al., 2021). This observation is another piece of evidence supporting the idea of a doubled Moho in the vicinity of the Moho depth offset, with the shallower Moho in the east extending above the deeper Moho in the west, as suggested by Luo et al. (2022). The observation of a doubled Moho has important implications on the origin of the Moho depth offset in southern New England. It favors a mechanism involving the overthrusting of the rifted Grenville crust and the Moretown terrane from the east onto undeformed Grenville in the west, along a reactivated Proterozoic rift fault (Luo et al., 2022). At the other end of the profile, the sin(baz) and cos(baz) signals associated with Feature #5 suggest that the Moho beneath the eastern portion of the array dips not only to the east but also to the south. This observation points to locally thicker crust to the southwest of the SEISConn array in central Rhode Island, which was also hinted at in previous larger-scale RF studies (e.g., Li et al., 2018).

Using the scattered wavefield migration technique, Luo et al. (2022) also identified a prominent west-dipping interface in the upper mantle cutting across the entire profile that represents an isotropic velocity increase with depth. This interface was interpreted as the Moho of a relict slab from a Paleozoic subduction event that was retained in the lithosphere after the deeper part had broken off. This feature is substantially weaker and less coherent in the traditional RF analysis (Luo et al., 2021), which can be attributed to a combination of the destructive interference between the top of the slab and the slab Moho as well as the possible partial eclogitization of the oceanic crust. Here, in the harmonically decomposed RFs, we can only glean the presence of this feature (Feature #6) in the constant RF profile (particularly in the higher frequency image in Figure 7). The higher-order harmonic signals associated with dipping interfaces are usually much smaller than the constant signals (e.g., Features #4 and #5). The velocity contrast across this dipping interface is likely too small for its harmonic signals to stand out and be distinguishable from other signals and background noise. This dipping interface may be associated with the formation of an anisotropic zone above it (Feature #7), which will be further discussed in Section 4.3.

4.2. Crustal Anisotropy

Our Bayesian inversion results (Figure 5) suggest that anisotropy with a steeply west-plunging slow axis may be present in the strata of the Hartford Basin, one of the abandoned rift basins created during the Mesozoic rifting (Hubert et al., 1992; Withjack et al., 2012). The Hartford Basin is a half-graben with a well-developed eastern border fault, and the layers within the basin are generally tilted to the east (Wise, 1992). The observed anisotropy can be explained by layering in the Hartford Basin. Fine layering in sedimentary basins can result in TI anisotropy with a layer-perpendicular axis of symmetry when layer thicknesses are smaller than the wavelength (Backus, 1962; Yan et al., 2016). The Hartford Basin consists of a series of layers with different seismic velocities including several basaltic flows (Philpotts & Martello, 1986). The thicknesses of those layers are much less than the wavelengths used in the RF analysis, which are on the order of kilometers, so they can result in anisotropic behaviors in RFs with a symmetry axis perpendicular to the layers. The observation of a generally west-plunging axis is consistent with the east-dipping layers in the Hartford Basin. The plunge has a mean value of 56° from the Bayesian inversion (Figure 5), which is in striking agreement with the ~55° dip of the eastern border fault in Connecticut (Digman, 1950) and complements the 15–35° dip of the strata (Wise, 1992). Furthermore, a mean axis trend of 110° is also consistent with the NNE strike of the eastern border fault and the basin layers (Wise, 1992).

Beneath the sedimentary strata of the Hartford Basin, there is a prominent layer with strong anisotropy as well as sharp top and bottom interfaces. This feature, observed beneath stations CS05-CS09, has a thickness of 3-4 km and a lateral extent of ~50 km (Feature #1 in Figures 7 and 8). The layer likely has an east-plunging slow axis of anisotropy, with >50° plunge and >20% anisotropy, as investigated in detail beneath CS07 (Figures 4 and 5). Anisotropy observed in the shallow upper crust with low confining pressure may be dominated by aligned cracks whose orientations are controlled by the stress field (Almqvist & Mainprice, 2017). Cracks orthogonal to the maximum compressive axis will be preferentially closed, and the seismic waves will propagate faster along this direction (Crampin & Chastin, 2003; Crampin & Lovell, 1991). Consistent with this mechanism, the observed fast directions of upper crustal azimuthal anisotropy across the contiguous U.S. are generally aligned with the maximum horizontal compressive stress direction (Lin & Schmandt, 2014). However, with a lateral extent of only ~50 km, we suggest that Feature #1 is likely not associated with the present-day regional stress field in the upper crust beneath southern New England; furthermore, its plunging geometry is not what would be expected for an aligned crack mechanism.

We instead favor an alternative explanation for this feature, which involves foliated metamorphic rocks and the CPO of crustal minerals. The CPO of mica and/or amphibole is usually invoked as the major contribution to substantial seismic anisotropy observed in the mid to lower crust (Lloyd et al., 2009; Rudnick & Fountain, 1995; Shapiro et al., 2004). Feature #1 is at 0–12 km depth today, but was likely at greater depths in the past. Insights into the history of crustal thickening and subsequent erosion and exhumation beneath New England come from geological constraints. Specifically, Hillenbrand and Williams (2021) proposed the existence of a long-lived orogenic plateau in southern New England during the Acadian and Neoacadian orogeny, which collapsed due to reduced compressive stresses after the Neoacadian orogeny. At present, surface rocks in southern New England were at larger depths in the past and were exhumed to the surface due to erosion and crustal thinning during and after the collapse of the Acadian orogenic plateau. Near the locality of Feature #1, the paleodepths of surface rocks are around 16-22 km (Hillenbrand & Williams, 2021), so this anisotropic layer would have been at 16-34 km depth before the proposed plateau collapse. This layer likely consists of metamorphic rocks such as schist and gneiss that are widespread in the nearby Gneiss Dome belts, Connecticut Valley basin, and the Bronson Hill arc (Connecticut Geological Survey, 2013). These metamorphic rocks contain minerals such as mica, amphibole and quartz that can form CPO under ductile deformation in the mid and lower crust (Almqvist & Mainprice, 2017). A hot and weak, perhaps partially molten, mid-crust susceptible to ductile flow was proposed for the Acadian orogenic plateau (Hillenbrand et al., 2022), similar to the mid-crust today beneath the Tibetan Plateau (e.g., Nelson et al., 1996). The anisotropy caused by CPO of mica can be approximated by a TI with a slow axis of symmetry, but deviations from TI are observed for amphibole, sillimanite, and quartz (Ji et al., 2015; Ward et al., 2012). The observation of strong slow-axis anisotropy with minor unmodeled harmonic signals (Figures S1d and S26 in Supporting Information S1) suggests that Feature #1 can be almost entirely described by TI, which may imply that the anisotropy is dominated by CPO of mica. Furthermore, the cleavage plane of mica is usually aligned with the foliation, which can result in seismic anisotropy with a slow axis of symmetry that is perpendicular to the foliation plane and a fast plane parallel to the foliation plane, while amphibole and sillimanite typically have crystallographic axes aligned with the lineation, which instead result in a seismic fast axis of symmetry that is subparallel to the lineation and a slow plane perpendicular to the lineation (Sherrington et al., 2004). Feature #1's well-constrained slow axis of symmetry is therefore consistent with the anisotropy caused by CPO of mica.

The observed east-plunging slow axis of Feature #1 suggests generally west-dipping foliation planes in this anisotropic layer. This west-dipping fabric might be associated with the shear zone resulting from the terrane accretion episodes during the Salinic and/or Acadian orogenies, which likely had west-dipping polarity (Robinson et al., 1998; van Staal et al., 2009). In the case of simple shear near suture faults, metamorphic rocks can develop S-C fabric, with the S-plane forming at ~45° to the shear direction, and the C-plane parallel to the shear direction (Lister & Snoke, 1984; Lloyd et al., 2009). The anisotropic behavior of Feature #1 is likely controlled by west-dipping foliation planes, which are parallel to the shear direction during west-dipping terrane accretion. Therefore, this layer likely experienced high shear strain during the Paleozoic terrane accretion event and is thus dominated by the shear motion-parallel C-fabric. A similar structure has been observed in the southern Appalachians at the Alleghanian Laurentia-Gondwana suture (Hopper et al., 2017). The Paleozoic fossil fabric associated with Feature #1 has been preserved until today, but the shape of this originally west-dipping layer was likely distorted by Mesozoic rifting and the opening of the Hartford Basin, resulting in the observed zig-zag shape beneath stations CS05-CS09.

Beneath Feature #1, there is a more elongated anisotropic interface identified as Feature #3, which may correspond to the anisotropic mid-crustal layer inferred by Gao et al. (2020). This feature extends from ~ 5 km depth beneath CS13 to ~20 km depth beneath CS03, which corresponds to an Acadian paleodepth of 21-42 km (Hillenbrand & Williams, 2021). This anisotropic feature is also likely associated with fossil CPO of crustal minerals in foliated metamorphic rocks because microcracks will generally be closed at pressures above 200–250 MPa (Ji et al., 2013) or about 10 km depth. Different from Feature #1, Feature #3 has a north- or south-plunging axis, a much smaller amplitude (suggesting weaker anisotropy), and a missing paired top or bottom interface that accompanies the observed interface, which makes the identification of its exact anisotropic geometry and strength impossible. The generally north-south fabric can be plausibly explained by a pure shear scenario with orogen-parallel minimum compressive axis, associated with the orogen-parallel crustal flow in the mid or lower crust during Acadian orogenic plateau collapse (Hillenbrand et al., 2022). These types of fabrics are evident in exposed metamorphic rocks with orogen-parallel fabrics to the north of our study area in southern Vermont and central Massachusetts (Karabinos et al., 2010; Massey et al., 2017). We speculate that the observed interface might be the relatively sharp top interface of the crustal flow zone, perhaps representing the brittle-ductile transition in the thickened Acadian crust (Kohlstedt et al., 1995; Mainprice & Nicolas, 1989), while the bottom interface is more diffuse and thus not observed in the RFs. The observation that Feature #3 is generally west-dipping may then imply either higher paleoelevation and/or more erosion or exhumation in eastern Connecticut than to the west.

An alternative explanation of Feature #3 may involve channel flow and ductile extrusion of the Putnam-Nashoba terrane in southeastern New England (Severson, 2020). Channel flow and extrusion of partially molten weak mid-crustal materials have been proposed for a number of convergent tectonic settings, including the Himalayas, the Canadian Cordillera, and the U.S. Appalachians (e.g., Beaumont et al., 2001; Kuiper et al., 2006; Merschat et al., 2005). In southeastern New England, the Putnam-Nashoba terrane might have extruded to the surface via channel flow during the Acadian orogeny, supported by geological evidence including the presence of partially melted rocks between lower grade metamorphic rocks above and below, and a transition of shear movements from normal to increasingly reverse toward the southeast (Severson, 2020). Feature #3 can be traced to the surface at the station CS13, which is near the western edge of the Putnam-Nashoba terrane. It is possible that Feature #3's anisotropy arises from the CPO of crustal minerals in the flow channel of the Putnam-Nashoba terrane. However, it is difficult to reconcile the eastward flow/extrusion with the observed north- or south-plunging anisotropic axis of Feature #3.

4.3. Mantle Anisotropy

Continental scale shear wave splitting studies have reported generally E-W or NE-SW oriented fast axes in the northeastern US, which suggest major contributions from plate motion parallel shearing of the asthenospheric mantle (Long et al., 2016; Yang et al., 2017), with the fast axis of olivine aligned with the flow direction (Zhang & Karato, 1995). Nevertheless, regional scale variation in splitting parameters is observed beneath southern New

England, with consistent plate motion parallel ENE-WSW fast directions but decreasing delay times toward the eastern end of the SEISConn array (Lopes et al., 2020). From our RF harmonics, we can identify two anisotropic layers with NE-SW trending symmetry in the upper mantle whose characteristics can potentially account for the decreasing delay times observed in the splitting analysis. One is a well-defined layer in the lithosphere associated with Feature #7, and the other is a zone in the asthenospheric mantle sandwiched between Feature #8 and Feature #9.

Feature #7 corresponds to a \sim 20 km thick anisotropic layer in the lithospheric mantle with well-defined top and bottom interfaces in the sin(2*baz) term (Figure 8). The harmonic signals observed in the sin(2*baz) term can be explained by either a NE-SW fast axis or a NW-SE slow axis. Given that seismic anisotropy in the upper mantle is usually attributed to olivine CPO, which under the dry conditions associated with continental lithospheric mantle (Peslier et al., 2017) has a fast axis roughly aligned with the shear direction (Karato, 2008; Long & Silver, 2009), Feature #7 likely corresponds to a generally NE-SW fast axis, reflecting fossil olivine CPO induced by a past episode of NE-SW directed shear. This shear zone overlies the relict slab lodged in the lithosphere (Feature #6; Luo et al., 2022) and may be associated with the deformation during the subduction of this slab in the Paleozoic. As discussed in Luo et al. (2022), the timing of this subduction event is not well constrained; it could have been during the Late Devonian Neoacadian orogeny, which does not have a clear record in southern New England (Kuiper et al., 2017; van Staal et al., 2009), or during the Carboniferous-Permian Alleghanian closure of the Rheic Ocean, whose subduction polarity is controversial (Domeier & Torsvik, 2014; Hermes & Murray, 1988; Michard et al., 2010, 2023; Nance & Linnemann, 2008; Nance et al., 2012). The observation that the shear zone, which lies only ~ 15 km deeper than the continental Moho, smoothly undercuts the prominent Moho depth offset between CS03 and CS04 (Figure 7) may suggest that the formation of the shear zone, and thus the subduction event, postdated the formation of the Moho depth offset. Hillenbrand et al. (2021) proposed that the Moho offset was mostly formed during the collapse of the Acadian orogenic plateau at ca. 330-310 Ma; this timing may favor an Alleghanian subduction origin for the relict slab. Regardless of the timing of subduction, the NE-SW fast axis in the shear zone implies that subduction was considerably oblique to the margin, given the general NNE strike of the Appalachian orogen at this latitude. One potential complication, pointed out by van Staal and Zagorevski (2023), is that there are apparently no signs of modification of the apparent relict slab proposed by Luo et al. (2022) by Central Atlantic Magmatic Province (CAMP) volcanism (e.g., Marzoli et al., 2018). Our interpretation of the mantle lithospheric shear zone, which relies on this interpretation of the dipping slab-like structure, thus having some uncertainties and warrants further investigation. Signals in cos(baz) term inside this anisotropic layer suggest complicated variations of anisotropic geometry within Feature #7, which may indeed reflect modifications by younger processes such as CAMP volcanism.

Features #8 and #9 lie at asthenospheric mantle depths and represent two interfaces that can be observed beneath nearly the entire SEISConn array. Because of the dipping nature of Feature #9, there is a triangular "wedge" of the mantle sandwiched between these two interfaces (Figure 9). These two features manifest in different harmonic terms, which implies that they are not just the top and bottom interfaces of a single anisotropic layer; furthermore, there may be additional anisotropy above and/or below this wedge-shaped layer. Feature #9, with negative sin(2*baz) signals, could be generated by an anisotropic layer with horizontal NE-SW fast axis above the interface, while the region beneath the interface can be either isotropic or have a vertical axis of anisotropic symmetry; these possibilities cannot be distinguished by anisotropic RF analysis (or shear wave splitting analysis). Feature #8 has dominant signals in the cos(baz) term, so the region above Feature #8 is likely also anisotropic with a north- or south-plunging axis of symmetry.

One possible explanation for Feature #8 is that it corresponds to the lithosphere-asthenosphere-boundary (LAB). An 80–100 km deep LAB beneath southern New England is somewhat deeper than the estimates from several previous RF studies (e.g., Goldhagen et al., 2022; Hopper & Fischer, 2018) but more consistent with others (e.g., Abt et al., 2010; Rychert et al., 2007). The nature of the LAB, and how it manifests in seismic observations, is still imperfectly understood (e.g., Karato & Park, 2019). A transition from a plunging fast axis above a horizontal fast axis below effectively represents a decrease in radial anisotropy with depth, which can be a plausible origin of seismically observed LAB beneath continents (Rychert & Shearer, 2009). Therefore, we propose that Feature #8 represents the top of an asthenospheric flow zone, which may also correspond to the LAB beneath southern New England at 80–100 km depth. In this interpretation, the anisotropic aspects of Feature #8 reflect the contrast between anisotropy near the base of the lithospheric mantle with a plunging symmetry axis and anisotropy in the asthenosphere with a nearly horizontal symmetry axis parallel to the plate motion.

We further hypothesize that the plate motion-parallel asthenospheric flow beneath New England is modified by the presence of a vertical upwelling flow near the eastern end of the profile. In this interpretation, Feature #9 represents the sharp transition of the mantle flow direction from vertical (below the boundary) to horizontal (above the boundary), such that the plate motion parallel flow zone becomes thinner to the east. The sharp transition of the mantle flow direction suggested by the anisotropy contrast can be achieved by a small-scale or edge-driven convection cell (e.g., King & Anderson, 1998; King & Ritsema, 2000). A thinner zone of plate motion parallel flow would result in smaller path-integrated shear wave splitting due to olivine CPO in the zone, which explains the observed weaker SKS splitting beneath the eastern portion of the SEISConn array (Lopes et al., 2020). One possibility is that the proposed mantle upwelling is associated with the NAA centered beneath central New England, thought to be a region of buoyant present-day upwelling (e.g., Levin et al., 1995; Menke et al., 2016). Our inference of mantle upwelling associated with the NAA beneath southern New England does not necessarily negate a possible contribution from the Great Meteor hotspot (e.g., Eaton & Frederiksen, 2007; Tao et al., 2020) in the formation of the NAA.

5. Summary and Implications

The anisotropic RF analysis for the SEISConn array presented here reveals the presence of multiple dipping interfaces and anisotropic layers in the crust and upper mantle beneath southern New England. The harmonic decomposition technique separates RF signals according to their backazimuthal dependence, which can then be further explored either by Bayesian inversion, to obtain quantitative constraints on the seismic model, or by tracing signals with the same harmonic patterns across adjacent stations to resolve the shape and lateral extent of individual features. We are able to identify relatively shallow anisotropic and dipping features in the crust with various potential origins, including fine layering in the Hartford Basin (Hubert et al., 1992), east-dipping faults associated with the Taconic thrust belt (Stanley & Ratcliffe, 1985), as well as foliation of metamorphic rocks and frozen-in CPO of crustal minerals developed during Salinic or Acadian terrane accretion events (van Staal et al., 2009). A more elongated anisotropic interface at mid-crustal depths may record orogen-parallel crustal flow associated with the collapse of the Acadian orogenic plateau (Hillenbrand et al., 2022) or channel flow and ductile extrusion of the Putnam-Nashoba terrane (Severson, 2020). The backazimuth-dependent signals produced by the dipping Moho discontinuity are also well captured by the harmonic analysis. The extension of harmonic signals associated with the west-dipping shallow Moho across the Moho depth offset represents additional evidence for a previously inferred doubled Moho, supporting an overthrust model for the formation of the Moho depth offset (Luo et al., 2022). There is an anisotropic layer in the lithospheric mantle that may represent a shear zone above a proposed relict slab lodged in the lithosphere due to an episode of oblique, west-dipping subduction during the Alleghanian closure of the Rheic Ocean. This anisotropic layer might also reflect modifications by younger processes such as CAMP volcanism (Marzoli et al., 2018). In the asthenosphere, our results suggest plate motion parallel flow in the asthenosphere. This asthenospheric flow zone may be modified by mantle upwelling associated with the NAA beneath the eastern end of the array; this model can explain the previously observed eastward decrease of SKS splitting delay times (Lopes et al., 2020).

This study provides perhaps the most comprehensive and in-depth regional analysis of crustal anisotropy beneath eastern North America to date, which has generally been lacking compared to the extensive studies of mantle anisotropy (Fouch & Rondenay, 2006). Quantitative analysis of crustal anisotropy using Bayesian inversion constrains the detailed geometry of prominent anisotropic structures beneath a single station, while the more qualitative analysis based on the dense array data can help identify robust features with smaller amplitudes. The constraints that we obtained on layered mantle anisotropy complement, and provide further insights into previously obtained results from shear wave splitting. The depth resolution provided by RF analysis is particularly valuable in regions with complicated mantle flow patterns that cannot be fully resolved using splitting analyses, such as the upwelling flow we infer at the eastern end of the SEISConn line.

The workflow that we have developed for this study also has great potential to be applied to other dense arrays that cross different segments of the Appalachian orogen (or to other orogenic regions). The general alignment of fast splitting directions with the strike of the Appalachian Mountains in the central and southern Appalachians, reported in previous SKS splitting studies (Aragon et al., 2017; Long et al., 2016), suggests strong contributions from lithospheric deformation associated with the Appalachian orogenesis. Furthermore, Appalachian orogenesis is known for substantial along-strike variation, which is manifested in not only the surface geology

15252027, 2023, 12, Downloaded from https:/ //agupub onlinelibrary.wiley.com/doi/10.1029/2023GC011118 by Yale University, Wiley Online Library on [15/12/2023]. . See the Terms

Acknowledgments

The authors thank two anonymous reviewers for their constructive comments that helped improve the manuscript. Collection and analysis of SEISConn data were funded by Yale University and by the National Science Foundation via Grants EAR-1150722 and EAR-1800923. The authors thank the station hosts and the field personnel, including the participants in the Field Experiences for Science Teachers (FEST) program (Long, 2017), for the successful SEISConn experiment. The analysis and interpretation of the results benefit from the discussions with colleagues, including Jay Ague, Shunichiro Karato, Jeffrey Park, and the Yale seismology group. The authors also thank Jeffrey Park for providing the original versions of the MTC and harmonic decomposition codes.

(Hatcher, 2010; Hibbard & Karabinos, 2013) but also in the Moho geometry (Luo et al., 2023). Detailed information about layered lithospheric anisotropy along the Appalachians can further our understanding of deformation processes and their lateral variations during Appalachian orogenesis, and can potentially be extrapolated to shed light on other orogenic belts, including those that are active today.

Data Availability Statement

Data used in this study from the SEISConn deployment (Long, 2015) are archived and made publicly available by the IRIS Data Management Center (DMC), which is operated by the EarthScope Consortium. SEIS-Conn metadata are available at http://ds.iris.edu/mda/XP/?starttime=2015-01-01T00:00:00&endtime=2019-12-31T23:59:59. The PyRaysum program (Bloch & Audet, 2023) for the computation of synthetic seismograms is available at https://github.com/paudetseis/PyRaysum.

References

- Abt, D. L., Fischer, K. M., French, S. W., Ford, H. A., Yuan, H., & Romanowicz, B. (2010). North American lithosphere discontinuity structure imaged by Ps and Sp receiver functions. Journal of Geophysical Research, 115(B9), 2009JB00691. https://doi.org/10.1029/2009jb006914
- Imaged by Ps and Sp receiver functions. Journal of Geophysical Research, 115(B9), 2009JB00691. https://doi.org/10.1029/2009Jb006914
 Almqvist, B. S. G., & Mainprice, D. (2017). Seismic properties and anisotropy of the continental crust: Predictions based on mineral texture and rock microstructure. Reviews of Geophysics, 55(2), 367–433. https://doi.org/10.1002/2016RG000552
- Ando, C. J., Czuchra, B. L., Klemperer, S. L., Brown, L. D., Cheadle, M. J., Cook, F. A., et al. (1984). Crustal profile of mountain belt: COCORP deep seismic reflection profiling in New England Appalachians and implications for architecture of convergent mountain chains 1. AAPG Bulletin, 68(7), 819–837. https://doi.org/10.1306/ad461430-16f7-11d7-8645000102c1865d
- Aragon, J. C., Long, M. D., & Benoit, M. H. (2017). Lateral variations in SKS splitting across the MAGIC array, central Appalachians. Geochemistry, Geophysics, Geosystems, 18(11), 4136–4155. https://doi.org/10.1002/2017GC007169
- Babuska, V., & Cara, M. (1991). Seismic anisotropy in the Earth (Vol. 10). Springer Science & Business Media.
- Backus, G. E. (1962). Long-wave elastic anisotropy produced by horizontal layering. Journal of Geophysical Research, 67(11), 4427–4440. https://doi.org/10.1029/JZ067i011p04427
- Bar, N., Long, M. D., Wagner, L. S., Beck, S. L., Zandt, G., & Tavera, H. (2019). Receiver function analysis reveals layered anisotropy in the crust and upper mantle beneath southern Peru and northern Bolivia. *Tectonophysics*, 753, 93–110. https://doi.org/10.1016/j.tecto.2019.01.007 Barruol, G., & Mainprice, D. (1993). A quantitative evaluation of the contribution of crustal rocks to the shear-wave splitting of teleseismic SKS
- waves. Physics of the Earth and Planetary Interiors, 78(3), 281–300. https://doi.org/10.1016/0031-9201(93)90161-2 Beaumont, C., Jamieson, R. A., Nguven, M. H., & Lee, B. (2001). Himalavan tectonics explained by extrusion of a low-viscosity crustal channel
- Seaumont, C., Jameson, K. A., Nguyen, M. H., & Lee, B. (2001). Himalayan tectonics explained by extrusion of a low-viscosity crustal channel coupled to focused surface denudation. *Nature*, 414(6865), 738–742. https://doi.org/10.1038/414738a
- Bianchi, I., Park, J., Piana Agostinetti, N., & Levin, V. (2010). Mapping seismic anisotropy using harmonic decomposition of receiver functions: An application to northern Apennines, Italy. *Journal of Geophysical Research*, 115(B12), B12317. https://doi.org/10.1029/2009JB007061
- Bissig, F., Khan, A., Tauzin, B., Sossi, P. A., Munch, F. D., & Giardini, D. (2021). Multifrequency inversion of Ps and Sp receiver functions: Methodology and application to USArray data. *Journal of Geophysical Research: Solid Earth*, 126(2), e2020JB020350. https://doi. org/10.1029/2020JB020350
- Bloch, W., & Audet, P. (2023). PyRaysum: Software for modeling ray-theoretical plane body-wave propagation in dipping anisotropic media. *Seismica*, 2(1). https://doi.org/10.26443/seismica.v2i1.220
- Bodin, T., Leiva, J., Romanowicz, B., Maupin, V., & Yuan, H. (2016). Imaging anisotropic layering with Bayesian inversion of multiple data types. *Geophysical Journal International*, 206(1), 605–629. https://doi.org/10.1093/gji/ggw124
- Bodin, T., Yuan, H., & Romanowicz, B. (2013). Inversion of receiver functions without deconvolution—Application to the Indian craton. *Geophysical Journal International*, 196(2), 1025–1033. https://doi.org/10.1093/gji/ggt431
- Bogdanova, S. V., Pisarevsky, S. A., & Li, Z. X. (2009). Assembly and breakup of Rodinia (some results of IGCP project 440). Stratigraphy and Geological Correlation, 17(3), 259–274. https://doi.org/10.1134/S0869593809030022
- Chapman, M. (2003). Frequency-dependent anisotropy due to meso-scale fractures in the presence of equant porosity. *Geophysical Prospecting*, 51(5), 369–379. https://doi.org/10.1046/j.1365-2478.2003.00384.x
- Christensen, N. I., & Mooney, W. D. (1995). Seismic velocity structure and composition of the continental crust: A global view. Journal of Geophysical Research, 100(B6), 9761–9788. https://doi.org/10.1029/95JB00259
- Connecticut Geological Survey. (2013). Generalized bedrock geologic map of Connecticut.
- Crampin, S. (1981). A review of wave motion in anisotropic and cracked elastic-media. Wave Motion, 3(4), 343-391. https://doi.org/10.1016/0165-2125(81)90026-3
- Crampin, S., & Chastin, S. (2003). A review of shear wave splitting in the crack-critical crust. *Geophysical Journal International*, 155(1), 221–240. https://doi.org/10.1046/j.1365-246X.2003.02037.x
- Crampin, S., & Lovell, J. H. (1991). A decade of shear-wave splitting in the Earth's crust: What does it mean? What use can we make of it? And what should we do next? *Geophysical Journal International*, 107(3), 387–407. https://doi.org/10.1111/j.1365-246X.1991.tb01401.x
- Digman, R. E. (1950). An exposure of the Triassic eastern border fault in Connecticut. American Journal of Science, 248(1), 37-45. https://doi.org/10.2475/ajs.248.1.37
- Domeier, M., & Torsvik, T. H. (2014). Plate tectonics in the late Paleozoic. *Geoscience Frontiers*, 5(3), 303–350. https://doi.org/10.1016/j.gsf.2014.01.002
- Eaton, D. W., & Frederiksen, A. (2007). Seismic evidence for convection-driven motion of the North American plate. *Nature*, 446(7134), 428–431. https://doi.org/10.1038/nature05675
- Ford, H. A., Long, M. D., & Wirth, E. A. (2016). Midlithospheric discontinuities and complex anisotropic layering in the mantle lithosphere beneath the Wyoming and Superior Provinces. *Journal of Geophysical Research: Solid Earth*, 121(9), 6675–6697. https://doi. org/10.1002/2016JB012978
- Fouch, M. J., Fischer, K. M., Parmentier, E. M., Wysession, M. E., & Clarke, T. J. (2000). Shear wave splitting, continental keels, and patterns of mantle flow. *Journal of Geophysical Research*, 105(B3), 6255–6275. https://doi.org/10.1029/1990JB900372

applicable C1

- Fouch, M. J., & Rondenay, S. (2006). Seismic anisotropy beneath stable continental interiors. *Physics of the Earth and Planetary Interiors*, 158(2), 292–320. https://doi.org/10.1016/j.pepi.2006.03.024
- Frederiksen, A. W., & Bostock, M. G. (2000). Modelling teleseismic waves in dipping anisotropic structures. *Geophysical Journal International*, 141(2), 401–412. https://doi.org/10.1046/j.1365-246x.2000.00090.x
- Frothingham, M. G., Schulte-Pelkum, V., Mahan, K. H., Merschat, A. J., Mather, M., & Gomez, Z. C. (2022). Don't judge an orogen by its cover: Kinematics of the Appalachian décollement from seismic anisotropy. *Geology*, 50(11), 1306–1311. https://doi.org/10.1130/g50323.1
- Gao, H., Yang, X., Long, M. D., & Aragon, J. C. (2020). Seismic evidence for crustal modification beneath the Hartford rift basin in the northeastern United States. *Geophysical Research Letters*, 47(17), e2020GL089316. https://doi.org/10.1029/2020gl089316
- Goldhagen, G. B., Ford, H. A., & Long, M. D. (2022). Evidence for a lithospheric step and pervasive lithospheric thinning beneath southern New England, northeastern USA. *Geology*, 50(9), 1078–1082. https://doi.org/10.1130/g50133.1
- Gung, Y., Panning, M., & Romanowicz, B. (2003). Global anisotropy and the thickness of continents. *Nature*, 422(6933), 707–711. https://doi.org/10.1038/nature01559
- Hansen, L. N., Faccenda, M., & Warren, J. M. (2021). A review of mechanisms generating seismic anisotropy in the upper mantle. *Physics of the Earth and Planetary Interiors*, 313, 106662. https://doi.org/10.1016/j.pepi.2021.106662
- Hatcher, R. D. (2002). Alleghanian (Appalachian) orogeny, a product of zipper tectonics: Rotational transpressive continent-continent collision and closing of ancient oceans along irregular margins (Vol. 364, pp. 199–208). Special Paper-Geological Society of America.
- Hatcher, R. D. (2010). The Appalachian orogen: A brief summary. From Rodinia to Pangea: The lithotectonic record of the Appalachian region (Vol. 206, pp. 1–19). Geological Society of America Memoir.
- Haws, A. A., Long, M. D., & Luo, Y. (2023). Anisotropic structure of the normally-dipping and flat slab segments of the Alaska subduction zone: Insights from receiver function analysis. *Tectonophysics*, 868, 230112. https://doi.org/10.1016/j.tecto.2023.230112
- Hermes, O., & Murray, D. P. (1988). Middle Devonian to Permian plutonism and volcanism in the N American Appalachians. *Geological Society*, London, Special Publications, 38(1), 559–571. https://doi.org/10.1144/gsl.Sp.1988.038.01.38
- Hibbard, J., & Karabinos, P. (2013). Disparate paths in the geologic evolution of the northern and southern Appalachians: A case for inherited contrasting crustal/lithospheric substrates. *Geoscience Canada*, 40(4), 303–317. https://doi.org/10.12789/geocanj.2013.40.021
- Hibbard, J. P., Van Staal, C. R., & Rankin, D. W. (2007). A comparative analysis of pre-Silurian crustal building blocks of the northern and the southern Appalachian orogen. American Journal of Science, 307(1), 23–45. https://doi.org/10.2475/01.2007.02
- Hibbard, J. P., van Staal, C. R., Rankin, D. W., Tollo, R. P., Bartholomew, M. J., & Karabinos, P. M. (2010). Comparative analysis of the geological evolution of the northern and southern Appalachian orogen: Late Ordovician-Permian. In *From Rodinia to Pangea: The lithotectonic record* of the Appalachian region (Vol. 206, p. 0). Geological Society of America. https://doi.org/10.1130/2010.1206(03)
- Hillenbrand, I., Williams, M. L., Jercinovic, M. J., Heizler, M. T., & Tjapkes, D. J. (2022). Petrochronologic constraints on Paleozoic tectonics in southern New England. In S. J. Whitmeyer, M. L. Williams, D. A. Kellett, & B. Tikoff (Eds.), *Laurentia: Turning points in the evolution of a continent* (p. 0). Geological Society of America. https://doi.org/10.1130/2022.1220(25)
- Hillenbrand, I. W., & Williams, M. L. (2021). Paleozoic evolution of crustal thickness and elevation in the northern Appalachian orogen, USA. Geology, 49(8), 946–951. https://doi.org/10.1130/g48705.1
- Hillenbrand, I. W., Williams, M. L., Li, C., & Gao, H. (2021). Rise and fall of the Acadian altiplano: Evidence for a Paleozoic orogenic plateau in New England. *Earth and Planetary Science Letters*, 560, 116797. https://doi.org/10.1016/j.epsl.2021.116797
- Holtzman, B. K., Kohlstedt, D. L., Zimmerman, M. E., Heidelbach, F., Hiraga, T., & Hustoft, J. (2003). Melt segregation and strain partitioning: Implications for seismic anisotropy and mantle flow. *Science*, 301(5637), 1227–1230. https://doi.org/10.1126/science.1087132
- Hopper, E., & Fischer, K. M. (2018). The changing face of the lithosphere-asthenosphere boundary: Imaging continental scale patterns in upper mantle structure across the contiguous U.S. With Sp converted waves. *Geochemistry, Geophysics, Geosystems*, 19(8), 2593–2614. https://doi. org/10.1029/2018GC007476
- Hopper, E., Fischer, K. M., Wagner, L. S., & Hawman, R. B. (2017). Reconstructing the end of the Appalachian orogeny. *Geology*, 45(1), 15–18. https://doi.org/10.1130/g38453.1
- Hubert, J. F., Feshbach-Meriney, P. E., & Smith, M. A. (1992). The Triassic-Jurassic Hartford rift basin, Connecticut and Massachusetts: Evolution, sandstone diagenesis, and hydrocarbon history 1. AAPG Bulletin, 76(11), 1710–1734. https://doi.org/10.1306/ bdff8ab0-1718-11d7-8645000102c1865d
- Ji, S., Shao, T., Michibayashi, K., Long, C., Wang, Q., Kondo, Y., et al. (2013). A new calibration of seismic velocities, anisotropy, fabrics, and elastic moduli of amphibole-rich rocks. Journal of Geophysical Research: Solid Earth, 118(9), 4699–4728. https://doi.org/10.1002/jgrb.50352
- Ji, S., Shao, T., Michibayashi, K., Oya, S., Satsukawa, T., Wang, Q., et al. (2015). Magnitude and symmetry of seismic anisotropy in mica- and amphibole-bearing metamorphic rocks and implications for tectonic interpretation of seismic data from the southeast Tibetan Plateau. *Journal* of Geophysical Research: Solid Earth, 120(9), 6404–6430. https://doi.org/10.1002/2015JB012209
- Kaneshima, S., Ando, M., & Kimura, S. (1988). Evidence from shear-wave splitting for the restriction of seismic anisotropy to the upper crust. *Nature*, 335(6191), 627–629. https://doi.org/10.1038/335627a0
- Karabinos, P., Macdonald, F. A., & Crowley, J. L. (2017). Bridging the gap between the foreland and hinterland I: Geochronology and plate tectonic geometry of Ordovician magmatism and terrane accretion on the Laurentian margin of New England. American Journal of Science, 317(5), 515–554. https://doi.org/10.2475/05.2017.01
- Karabinos, P., Mygatt, E. S., Cook, S. M., Student, M., Tollo, R. P., Bartholomew, M. J., et al. (2010). Evidence for an orogen-parallel, normal-sense shear zone around the Chester dome, Vermont: A possible template for gneiss dome formation in the New England Appalachians, USA. In From Rodinia to Pangea: The lithotectonic record of the Appalachian region (Vol. 206, p. 0). Geological Society of America. https://doi.org/10.1130/2010.1206(09)
- Karato, S., & Park, J. (2019). On the origin of the upper mantle seismic discontinuities. In H. Yuan, & B. Romanowicz (Eds.), Lithospheric disconstinuities (pp. 5–34). American Geophysical Union.
- Karato, S.-I. (2008). Deformation of Earth materials: An introduction to the rheology of solid Earth. Cambridge University Press. https://doi. org/10.1017/CBO9780511804892
- Karato, S.-I., & Wu, P. (1993). Rheology of the upper mantle: A synthesis. Science, 260(5109), 771–778. https://doi.org/10.1126/ science.260.5109.771
- Kennett, B. L. N., & Engdahl, E. R. (1991). Traveltimes for global earthquake location and phase identification. Geophysical Journal International, 105(2), 429–465. https://doi.org/10.1111/j.1365-246X.1991.tb06724.x
- King, S. D., & Anderson, D. L. (1998). Edge-driven convection. Earth and Planetary Science Letters, 160(3), 289–296. https://doi.org/10.1016/ S0012-821X(98)00089-2
- King, S. D., & Ritsema, J. (2000). African hot spot volcanism: Small-scale convection in the upper mantle beneath cratons. Science, 290(5494), 1137–1140. https://doi.org/10.1126/science.290.5494.1137

- Kohlstedt, D. L., Evans, B., & Mackwell, S. J. (1995). Strength of the lithosphere: Constraints imposed by laboratory experiments. Journal of Geophysical Research, 100(B9), 17587–17602. https://doi.org/10.1029/95JB01460
- Kuiper, Y. D., Thompson, M. D., Barr, S. M., White, C. E., Hepburn, J. C., & Crowley, J. L. (2017). Detrital zircon evidence for Paleoproterozoic West African crust along the eastern North American continental margin, Georges Bank, offshore Massachusetts, USA. *Geology*, 45(9), 811–814. https://doi.org/10.1130/g39203.1
- Kuiper, Y. D., Williams, P. F., & Kruse, S. (2006). Possibility of channel flow in the southern Canadian Cordillera: A new approach to explain existing data. *Geological Society, London, Special Publications*, 268(1), 589–611. https://doi.org/10.1144/GSL.SP.2006.268.01.27
- Langston, C. A. (1979). Structure under Mount Rainier, Washington, inferred from teleseismic body waves. Journal of Geophysical Research, 84(B9), 4749–4762. https://doi.org/10.1029/JB084iB09p04749
- Levin, V., Lerner-Lam, A., & Menke, W. (1995). Anomalous mantle structure at the Proterozoic-Paleozoic boundary in northeastern US. Geophysical Research Letters, 22(2), 121–124. https://doi.org/10.1029/94GL02693
- Levin, V., Long, M. D., Skryzalin, P., Li, Y., & López, I. (2018). Seismic evidence for a recently formed mantle upwelling beneath New England. *Geology*, 46(1), 87–90. https://doi.org/10.1130/g39641.1
- Levin, V., & Park, J. (1997). P-SH conversions in a flat-layered medium with anisotropy of arbitrary orientation. Geophysical Journal International, 131(2), 253–266. https://doi.org/10.1111/j.1365-246X.1997.tb01220.x
- Li, A., Forsyth, D. W., & Fischer, K. M. (2003). Shear velocity structure and azimuthal anisotropy beneath eastern North America from Rayleigh wave inversion. *Journal of Geophysical Research*, 108(B8), 2362. https://doi.org/10.1029/2002JB002259
- Li, C., Gao, H., Williams, M. L., & Levin, V. (2018). Crustal thickness variation in the northern Appalachian mountains: Implications for the geometry of 3-D tectonic boundaries within the crust. *Geophysical Research Letters*, 45(12), 6061–6070. https://doi.org/10.1029/2018GL078777
- Li, Y., Levin, V., Nikulin, A., & Chen, X. (2021). Systematic mapping of upper mantle seismic discontinuities beneath northeastern North America. Geochemistry, Geophysics, Geosystems, 22(7), e2021GC009710. https://doi.org/10.1029/2021GC009710
- Lin, F.-C., & Schmandt, B. (2014). Upper crustal azimuthal anisotropy across the contiguous U.S. determined by Rayleigh wave ellipticity. Geophysical Research Letters, 41(23), 8301–8307. https://doi.org/10.1002/2014GL062362
- Link, F., Reiss, M. C., & Rümpker, G. (2022). An automatized XKS-splitting procedure for large data sets: Extension package for SplitRacer and application to the USArray. *Computers & Geosciences*, 158, 104961. https://doi.org/10.1016/j.cageo.2021.104961
- Link, F., Rümpker, G., & Kaviani, A. (2020). Simultaneous inversion for crustal thickness and anisotropy by multiphase splitting analysis of receiver functions. *Geophysical Journal International*, 223(3), 2009–2026. https://doi.org/10.1093/gji/ggaa435
- Lister, G. S., & Snoke, A. W. (1984). S-C mylonites. *Journal of Structural Geology*, 6(6), 617–638. https://doi.org/10.1016/0191-8141(84)90001-4 Liu, Z., Park, J., & Rye, D. M. (2015). Crustal anisotropy in northeastern Tibetan Plateau inferred from receiver functions: Rock textures caused
- by metamorphic fluids and lower crust flow? *Tectonophysics*, 661, 66–80. https://doi.org/10.1016/j.tecto.2015.08.006 Lloyd, G. E., Butler, R. W. H., Casey, M., & Mainprice, D. (2009). Mica, deformation fabrics and the seismic properties of the continental crust. *Earth and Planetary Science Letters*, 288(1), 320–328. https://doi.org/10.1016/j.epsl.2009.09.035
- Long, M. D. (2015). Seismic experiment for imaging structure beneath Connecticut [Dataset]. International Federation of Digital Seismograph Networks. https://doi.org/10.7914/SN/XP_2015
- Long, M. D. (2017). The field experiences for science teachers (FEST) program: Involving Connecticut high school science teachers in field seismology. Seismological Research Letters, 88(2A), 421–429. https://doi.org/10.1785/0220160162
- Long, M. D., & Aragon, J. C. (2020). Probing the structure of the crust and mantle lithosphere beneath the southern New England Appalachians via the SEISConn deployment. Seismological Research Letters, 91(5), 2976–2986. https://doi.org/10.1785/0220200163
- Long, M. D., Ford, H. A., Abrahams, L., & Wirth, E. A. (2017). The seismic signature of lithospheric deformation beneath eastern North America due to Grenville and Appalachian orogenesis. *Lithosphere*, 9(6), 987–1001. https://doi.org/10.1130/1660.1
- Long, M. D., Jackson, K. G., & McNamara, J. F. (2016). SKS splitting beneath Transportable Array stations in eastern North America and the signature of past lithospheric deformation. Geochemistry, Geophysics, Geosystems, 17(1), 2–15. https://doi.org/10.1002/2015GC006088
- Long, M. D., & Silver, P. G. (2009). Shear wave splitting and mantle anisotropy: Measurements, interpretations, and new directions. Surveys in Geophysics, 30(4), 407–461. https://doi.org/10.1007/s10712-009-9075-1
- Lopes, E., Long, M. D., Karabinos, P., & Aragon, J. C. (2020). SKS splitting and upper mantle anisotropy beneath the southern New England Appalachians: Constraints from the dense SEISConn array. *Geochemistry, Geophysics, Geosystems*, 21(12), e2020GC009401. https://doi. org/10.1029/2020GC009401
- Luo, Y., Long, M. D., Karabinos, P., Kuiper, Y. D., Rondenay, S., Aragon, J. C., et al. (2021). High-resolution Ps receiver function imaging of the crust and mantle lithosphere beneath southern New England and tectonic implications. *Journal of Geophysical Research: Solid Earth*, 126(7), e2021JB022170. https://doi.org/10.1029/2021JB022170
- Luo, Y., Long, M. D., Karabinos, P., Rondenay, S., & Masis Arce, R. (2023). First-order transition in Appalachian orogenic processes revealed by along-strike variation of the Moho geometry. *Journal of Geophysical Research: Solid Earth*. https://doi.org/10.1029/2023JB027024
- Luo, Y., Long, M. D., Rondenay, S., Karabinos, P., & Kuiper, Y. D. (2022). Wavefield migration imaging of Moho geometry and upper mantle structure beneath southern New England. *Geophysical Research Letters*, 49(13), e2022GL099013. https://doi.org/10.1029/2022GL099013
- Macdonald, F. A., Ryan-Davis, J., Coish, R. A., Crowley, J. L., & Karabinos, P. (2014). A newly identified Gondwanan terrane in the northern Appalachian Mountains: Implications for the Taconic orogeny and closure of the Iapetus Ocean. *Geology*, 42(6), 539–542. https://doi. org/10.1130/g35659.1
- Mainprice, D., & Nicolas, A. (1989). Development of shape and lattice preferred orientations: Application to the seismic anisotropy of the lower crust. Journal of Structural Geology, 11(1), 175–189. https://doi.org/10.1016/0191-8141(89)90042-4
- Marzoli, A., Callegaro, S., Dal Corso, J., Davies, J. H. F. L., Chiaradia, M., Youbi, N., et al. (2018). The central Atlantic magmatic province (CAMP): A review. In L. H. Tanner (Ed.), *The late Triassic world: Earth in a time of transition* (pp. 91–125). Springer International Publishing. https://doi.org/10.1007/978-3-319-68009-5_4
- Massey, M. A., Moecher, D. P., Walker, T. B., O'Brien, T. M., & Rohrer, L. P. (2017). The role and extent of dextral transpression and lateral escape on the post-Acadian tectonic evolution of south-central New England. *American Journal of Science*, 317(1), 34–94. https://doi.org/10.2475/01.2017.02
- Menke, W., Skryzalin, P., Levin, V., Harper, T., Darbyshire, F., & Dong, T. (2016). The northern Appalachian anomaly: A modern asthenospheric upwelling. *Geophysical Research Letters*, 43(19), 10173–10179. https://doi.org/10.1002/2016GL070918
- Merschat, A. J., Hatcher, R. D., & Davis, T. L. (2005). The northern Inner Piedmont, southern Appalachians, USA: Kinematics of transpression and SW-directed mid-crustal flow. Journal of Structural Geology, 27(7), 1252–1281. https://doi.org/10.1016/j.jsg.2004.08.005
- Michard, A., Driouch, Y., Kuiper, Y. D., Caby, R., Farah, A., Ouanaimi, H., et al. (2023). The Variscan belts of North-West Africa: An African legacy to the Wilson Cycle concept. *Journal of African Earth Sciences*, 208, 105042. https://doi.org/10.1016/j.jafrearsci.2023.105042

- Michard, A., Soulaimani, A., Hoepffner, C., Ouanaimi, H., Baidder, L., Rjimati, E. C., & Saddiqi, O. (2010). The south-western branch of the Variscan Belt: Evidence from Morocco. *Tectonophysics*, 492(1), 1–24. https://doi.org/10.1016/j.tecto.2010.05.021
- Nance, R. D., Gutiérrez-Alonso, G., Keppie, J. D., Linnemann, U., Murphy, J. B., Quesada, C., et al. (2012). A brief history of the Rheic Ocean. Geoscience Frontiers, 3(2), 125–135. https://doi.org/10.1016/j.gsf.2011.11.008
- Nance, R. D., & Linnemann, U. (2008). The Rheic Ocean: Origin, evolution, and significance. *Geological Society of America Today*, 18(12), 4–12. https://doi.org/10.1130/gsatg24a.1
- Nelson, K. D., Zhao, W., Brown, L. D., Kuo, J., Che, J., Liu, X., et al. (1996). Partially molten middle crust beneath southern Tibet: Synthesis of project INDEPTH results. Science, 274(5293), 1684–1688. https://doi.org/10.1126/science.274.5293.1684
- Olugboji, T. M., Park, J., Karato, S., & Shinohara, M. (2016). Nature of the seismic lithosphere-asthenosphere boundary within normal oceanic mantle from high-resolution receiver functions. *Geochemistry, Geophysics, Geosystems*, 17(4), 1265–1282. https://doi. org/10.1002/2015gc006214
- Park, J., & Levin, V. (2000). Receiver functions from multiple-taper spectral correlation estimates. Bulletin of the Seismological Society of America, 90(6), 1507–1520. https://doi.org/10.1785/0119990122
- Park, J., & Levin, V. (2002). Seismic anisotropy: Tracing plate dynamics in the mantle. Science, 296(5567), 485–489. https://doi.org/10.1126/ science.1067319
- Park, J., & Levin, V. (2016a). Anisotropic shear zones revealed by backazimuthal harmonics of teleseismic receiver functions. *Geophysical Journal International*, 207(2), 1216–1243. https://doi.org/10.1093/gji/ggw323
- Park, J., & Levin, V. (2016b). Statistics and frequency-domain moveout for multiple-taper receiver functions. *Geophysical Journal International*, 207(1), 512–527. https://doi.org/10.1093/gji/ggw291
- Peslier, A. H., Schönbächler, M., Busemann, H., & Karato, S.-I. (2017). Water in the Earth's interior: Distribution and origin. Space Science Reviews, 212(1), 743–810. https://doi.org/10.1007/s11214-017-0387-z
- Philpotts, A. R., & Martello, A. (1986). Diabase feeder dikes for the Mesozoic basalts in southern New England. American Journal of Science, 286(2), 105–126. https://doi.org/10.2475/ajs.286.2.105
- Rivers, T. (1997). Lithotectonic elements of the Grenville Province: Review and tectonic implications. Precambrian Research, 86(3), 117–154. https://doi.org/10.1016/S0301-9268(97)00038-7
- Robinson, P., Tucker, R. D., Bradley, D., Berry, H. N., & Osberg, P. H. (1998). Paleozoic orogens in New England, USA. GFF, 120(2), 119–148. https://doi.org/10.1080/11035899801202119
- Rondenay, S. (2009). Upper mantle imaging with array recordings of converted and scattered teleseismic waves. *Surveys in Geophysics*, 30(4–5), 377–405. https://doi.org/10.1007/s10712-009-9071-5
- Rudnick, R. L., & Fountain, D. M. (1995). Nature and composition of the continental crust: A lower crustal perspective. *Reviews of Geophysics*, 33(3), 267–309. https://doi.org/10.1029/95RG01302
- Rychert, C. A., Rondenay, S., & Fischer, K. M. (2007). P-to-S and S-to-P imaging of a sharp lithosphere-asthenosphere boundary beneath eastern North America. *Journal of Geophysical Research*, 112(B8), B08314. https://doi.org/10.1029/2006JB004619
- Rychert, C. A., & Shearer, P. M. (2009). A global view of the lithosphere-asthenosphere boundary. Science, 324(5926), 495–498. https://doi. org/10.1126/science.1169754
- Savage, M. K. (1999). Seismic anisotropy and mantle deformation: What have we learned from shear wave splitting? *Reviews of Geophysics*, 37(1), 65–106. https://doi.org/10.1029/98RG02075
- Schulte-Pelkum, V., & Mahan, K. H. (2014a). Imaging faults and shear zones using receiver functions. Pure and Applied Geophysics, 171(11), 2967–2991. https://doi.org/10.1007/s00024-014-0853-4
- Schulte-Pelkum, V., & Mahan, K. H. (2014b). A method for mapping crustal deformation and anisotropy with receiver functions and first results from USArray. Earth and Planetary Science Letters, 402, 221–233. https://doi.org/10.1016/j.epsl.2014.01.050
- Severson, A. R. (2020). Across- and along-strike structural and geochronological variations of the Nashoba-Putnam and Avalon terranes, eastern Massachusetts, Connecticut, and Rhode Island, southeastern New England Appalachians. In *Colorado school of mines*. Arthur Lakes Library.
- Shapiro, N. M., Ritzwoller, M. H., Molnar, P., & Levin, V. (2004). Thinning and flow of Tibetan crust constrained by seismic anisotropy. *Science*, 305(5681), 233–236. https://doi.org/10.1126/science.1098276
- Sherrington, H. F., Zandt, G., & Frederiksen, A. (2004). Crustal fabric in the Tibetan Plateau based on waveform inversions for seismic anisotropy parameters. Journal of Geophysical Research, 109(B2), B02312. https://doi.org/10.1029/2002JB002345
- Shiomi, K., & Park, J. (2008). Structural features of the subducting slab beneath the Kii Peninsula, central Japan: Seismic evidence of slab segmentation, dehydration, and anisotropy. Journal of Geophysical Research, 113(B10), B10318. https://doi.org/10.1029/2007JB005535
- Stanley, R. S., & Ratcliffe, N. M. (1985). Tectonic synthesis of the Taconian orogeny in western New England. GSA Bulletin, 96(10), 1227–1250. https://doi.org/10.1130/0016-7606(1985)96<1227:Tsotto>2.0.Co;2
- Tao, Z., Li, A., & Fischer, K. M. (2020). Hotspot signatures at the North American passive margin. Geology, 49(5), 525–530. https://doi.org/10.1130/g47994.1
- van Staal, C. R., Whalen, J. B., Valverde-Vaquero, P., Zagorevski, A., & Rogers, N. (2009). Pre-Carboniferous, episodic accretion-related, orogenesis along the Laurentian margin of the northern Appalachians. *Geological Society, London, Special Publications*, 327(1), 271–316. https:// doi.org/10.1144/sp327.13
- van Staal, C. R., & Zagorevski, A. (2023). A note on the timing and nature of the Moho geometry and upper mantle structure beneath southern New England. Geophysical Research Letters, 50(6), e2022GL102057. https://doi.org/10.1029/2022GL102057
- Ward, D., Mahan, K., & Schulte-Pelkum, V. (2012). Roles of quartz and mica in seismic anisotropy of mylonites. *Geophysical Journal Interna*tional, 190(2), 1123–1134. https://doi.org/10.1111/j.1365-246X.2012.05528.x
- White-Gaynor, A. L., & Nyblade, A. A. (2017). Shear wave splitting across the Mid-Atlantic region of North America: A fossil anisotropy interpretation. *Geology*, 45(6), 555–558. https://doi.org/10.1130/g38794.1
- Wirth, E. A., & Long, M. D. (2014). A contrast in anisotropy across mid-lithospheric discontinuities beneath the central United States—A relic of craton formation. *Geology*, 42(10), 851–854. https://doi.org/10.1130/g35804.1
- Wirth, E. A., Long, M. D., & Moriarty, J. C. (2016). A Markov chain Monte Carlo with Gibbs sampling approach to anisotropic receiver function forward modeling. *Geophysical Journal International*, 208(1), 10–23. https://doi.org/10.1093/gji/ggw383
- Wise, D. U. (1992). Dip domain method applied to the Mesozoic Connecticut Valley rift basins. *Tectonics*, 11(6), 1357–1368. https://doi.org/10.1029/92TC01103
- Withjack, M. O., Schlische, R. W., & Olsen, P. E. (2012). Development of the passive margin of eastern North America: Mesozoic rifting, igneous activity, and breakup. Regional Geology and Tectonics: Phanerozoic Rift Systems and Sedimentary Basins, 1, 301.
- Yan, F., Han, D.-H., Sil, S., & Chen, X.-L. (2016). Analysis of seismic anisotropy parameters for sedimentary strata. *Geophysics*, 81(5), D495–D502. https://doi.org/10.1190/geo2016-0062.1

- Yang, B. B., Liu, Y., Dahm, H., Liu, K. H., & Gao, S. S. (2017). Seismic azimuthal anisotropy beneath the eastern United States and its geodynamic implications. *Geophysical Research Letters*, 44(6), 2670–2678. https://doi.org/10.1002/2016GL071227
- Yuan, H., & Levin, V. (2014). Stratified seismic anisotropy and the lithosphere-asthenosphere boundary beneath eastern North America. Journal of Geophysical Research: Solid Earth, 119(4), 3096–3114. https://doi.org/10.1002/2013JB010785
- Zhang, S., & Karato, S.-I. (1995). Lattice preferred orientation of olivine aggregates deformed in simple shear. *Nature*, 375(6534), 774–777. https://doi.org/10.1038/375774a0

References From the Supporting Information

- Çakır, Ö., Erduran, M., Çınar, H., & Yılmaztürk, A. (2000). Forward modelling receiver functions for crustal structure beneath station TBZ (Trabzon, Turkey). Geophysical Journal International, 140(2), 341–356. https://doi.org/10.1046/j.1365-246x.2000.00023.x
- Chevrot, S., & Girardin, N. (2000). On the detection and identification of converted and reflected phases from receiver functions. *Geophysical Journal International*, 141(3), 801–808. https://doi.org/10.1046/j.1365-246x.2000.00139.x
- Everitt, B. S., Landau, S., Leese, M., & Stahl, D. (2011). Cluster analysis. Wiley. https://doi.org/10.1002/9780470977811
- Hastings, W. K. (1970). Monte Carlo sampling methods using Markov chains and their applications. *Biometrika*, 57(1), 97–109. https://doi.org/10.2307/2334940
- Park, J., & Levin, V. (2016). Statistics and frequency-domain moveout for multiple-taper receiver functions. *Geophysical Journal International*, 207(1), 512–527. https://doi.org/10.1093/gji/ggw291
- Rondenay, S., Spieker, K., Sawade, L., Halpaap, F., & Farestveit, M. (2017). GLImER: A new global database of teleseismic receiver functions for imaging Earth structure. Seismological Research Letters, 88(1), 39–48. https://doi.org/10.1785/0220160111